

# Collective Agreement

between

EMC Emergency Medical Care Incorporated

and the

International Union of Operating Engineers, Local 968B

(April 1, 2005 - March 31,2008)

SMSS\\1051363.15

123/3/63)

# TABLE OF CONTENTS

# PREAMBLE

ARTICLE	1 - DEFINITIONS
1.01	Definitions
1.02	Gender
ARTICLE	2 - MANAGEMENT RIGHTS
2.01	Management Rights
2.02	Consistent Application
2.03	Identification of Supervisory Officials
	1 ,
ARTICLE	3 - RECOGNITION
3.01	Bargaining Unit Recognition
3.02	Mutual Agreements
3.03	Bargaining Unit Work - Supervisors
ARTICLE	4 - APPLICATION
ARTICLE	5 - FUTURE LEGISLATION
ADTICI E	6 - NO DISCRIMINATION
6.01	No Discrimination for Union Activity
6.02	No Discrimination - Human Rights
6.03	Sexual and Personal Harassment
0.03	Sexual and Leisonal Harassment
ARTICI F	7 - RIGHTS AND PROHIBITIONS
7.01	No Lockout or Strike
7.02	No Sanction of Strike
7.03	Picket Line or Other Strike Activity
1.05	Tienet Elile of Other Birthe Tientytey
ARTICLE	8 - CHECK OFF
8.01	Deduction
8.02	Notification of Deduction

# 8.05 Income Tax Receipt for Dues

Remittance of Union Dues

ARTICLE	9 - UNION REPRESENTATION AND AFFAIRS
9.01	Appointment of Stewards

Indemnification

9.02 Steward's Duties 9.03 Union Officials

9.04 Negotiating Committee

# ARTICLE 10 - EMPLOYEE INFORMATION

10.01 Acquainting New Employees 10.02 Printing of Collective Agreement

SMSS\\1051363 15

8.03 8.04

10.03	Bulletin Boards
10.04	EHS Provincial Medical Policies and Procedures Manual
10.05	Job Descriptions
ARTICLE	11 - GRIEVANCE PROCEDURE
11.01	Complaints
11.02	
	Steps in Grievance Procedure
11.03	Suspension or Discharge Grievances
11.04	Union Policy Grievance
11.05	Employer Grievance
11.06	Time Limits
11.07	Grievance Hearings
ADTICI E	12 - ARBITRATION PROCEDURE
12.0I	Reference to Arbitration
12.02	Single Arbitrator Unless Board Requested
12.03	Single Arbitrator
12.04	Arbitration Board
12.05	Authority of Arbitrator/Board
12.06	Hearing and Decision
12.07	Fees and Disbursements of Arbitrator/Board
12.08	Time Limits
ARTICLE	13 - RESIGNATION
13,01	Notice of Resignation
13.02	Failure to Give Notice
13.03	Withdrawal of Resignation
10100	The state of the s
ARTICLE	14 - PERFORMANCE APPRAISALS
ARTICLE	15 - TIME OFF FOR UNION BUSINESS
15.01	Leave for Union Business
15.02	Notification to Employer
15.03	Annual Meeting
15.04	Salary Continuance
15.05	Arbitrations and Joint Consultation
15.06	Leave of Absence for Full-Time Union President
15.07	No Loss of Service, Seniority or Benefits
13.07	No Loss of Service, Semonty of Benefits
ARTICI F	16 - APPOINTMENT
16.01	Probation
16.02	Dismissal of Probationary Employee
16.03	Union Notification
ARTICLE	17 - DISCHARGE, SUSPENSION AND DISCIPLINE
17.01	Just Cause
17.02	Notification
17.02	
17.03	Record of Disciplinary Action
SMSS\\1051363	315

17.04	Employee Access to Human Resources File
17.05	Right to Have Steward Present
ARTICLE	I8 - SENIORITY
18.01	Seniority Defined
18.02	Posting of Seniority Lists
18.03	Loss of Seniority
18.04	Service
18.05	Service Credits
ARTICLE	19 - HOLIDAYS
19.01	Holidays
19.02	Exception
19.03	Holiday Coinciding With Paid Leave
19.04	Compensation for Work on a Holiday
19.05	Christmas Day or New Years Off
	••
	20 - JOB POSTING
20.01	Job Posting
20.02	Job Postings (General)
20.03	Filling Position for a Field Paramedic
20.04	Promotion
20.05	Temporary Transfers
20.06	Reassignments
20.07	Secondments
20.08	Trial Period
20.09	Non-Bargaining Unit Vacant Position
20.10	No Avoidance
ADTICI E	21 IOD SECUDITY
	21 - JOB SECURITY Joint Committee on Labour Relations
21.01	Definition
21.02	Introduction
21.03 21.04	Notice to Union
21.05	Training
21.06	Application
21.07	Notice of Layoff
21.08	Pay in Lieu of Notice
21.09	Placement
21.10	Displacement Procedures
21.11	Recall Procedures
21.12	Termination of Recall Rights
ARTICLE	22 - JOINT CONSULTATION
22.01	Labour Management Committees
22.02	Terms of Reference

ARTICLE 23.01 23.02 23.03	23 - HEALTH AND SAFETY Occupational Health and Safety Act Occupational Health and Safety Committee(s) Alcohol, Drug and Gambling Dependency
ARTICLE 24.01 24.02	24 - UNIFORMS Uniforms and Equipment Safety Boots
25.01 25.02 25.03 25.04 25.05	25 - GENERAL CONDITIONS Rate of Pay During Accident Investigation Crew Quarters Drivers License EHS Registration Deep Cleaning of Vehicles
	26 - PAY PROVISION
26.01	Rate of Pay
26.02	Rate of Pay Upon Appointment
26.03	Exception
26.04	Rate of Pay Upon Reclassification
26.05	Anniversary Date
26.06	Salary Increments
26.07	Notice of Withheld Increment
26.08	Granting of Withheld Increment
26.09	Premium for Senior Operations Paramedics Rate of Pay Other Than Regular Rate
26.10	Rate of Fay Other Than Regular Rate
ARTICLE	27 • HOURS OF WORK
27.01	No Guarantee of Hours
27.02	Regular Work Week
27.03	Shifts
27.04	Breaks and Allowances
27.05	Posted Schedule
27.06	Compensation for Partial Shift
27.07	Consecutive Days Off
27.08	Stand-By
	28 - OVERTIME Overtime
28.01	
28.02	Call-In System
28.03	Premium Pay
28.04	Overtime Eligibility Computation of Overtime
28.05	Form of Compensation/Banking Excess Hours
28.06	Call-In
28.07	
28.08 28.09	Call-In System Overtime Meal Allowance
20,07	

# 28.10 Overtime Restriction

ARTICLE	29 - LEAVES OF ABSENCE	
29.01	Personal Leave	
29.02	Sick Leave	
29.03	Other Uses for Sick Leave	
(a)	Sick Leave for Family Illness	
(b)	Leave for Medical and Dental Appointments	
(c)	Leave for Birth or Adoption of Child	
(d)	Leave for Emergency	
29.04	Summary/Clarification	
29.05	Leave for Family Illness	
29.06	Proration of Sick Leave	
29.07	Court Leave	
29.08	Pregnancy Leave	
29.09	Pregnancy Leave Allowance	
29.10	Parental Leave	
29.11	Adoption Leave	
29.12	Adoption Leave Allowance	
29.13	Interview Leave	
29.14	Bereavement Leave	
ARTICLE	30 - VACATIONS	
30.01	Vacations	
30.02	Vacation Year	
30.03	Vacation Requests	
30.04	Vacation Scheduling	
30.05	Payout of Vacation Entitlements	
30.06	Illness During Vacation	
30.07	Bereavement Leave During Vacation	
30.08	Vacation Carry-Over	
30.09	Vacation Records	
30.10	Recall from Vacation	
ARTICLE	31 - EMPLOYEE BENEFITS	
31.01	Pension	
31.02	Group Life Insurance Plan	
31.03	Extended Health Benefit Plan (Medical and Dental)	
31,04	Short Term Disability and Long Term Disability	
31.05	Regular Part-Time Employees	
31.06	Benefits Committee	
31.07	Employer's Responsibility	
ARTICI F	E 32 - SEVERANCE PAY	
32.01	Severance Pay	
32.02	Outstanding Debts Upon Death	
52.02	Catalanding Deets Open Death	

ARTICLE 33 - EMPLOYER LIABILITY

33.01 Legal Expenses

ARTICLE 34 - AMENDMENT

ARTICLE 35 – CORRESPONDENCE

ARTICLE 36 - TERM OF AGREEMENT

36.01 Term

36.02 Retroactivity

SCHEDULE A - SALARY SCALE

SCHEDULE B - ADMINISTRATIVE AREAS

MEMORANDUM OF AGREEMENT #1 (RE ARTICLE 18 - SENIORITY)

MEMORANDUM OF AGREEMENT #2 (RE GRANDFATHERING OF SERVICE CREDITS)

MEMORANDUM OF AGREEMENT #3 (RE POLICY OF SHIFT SWITCHES)

### **PREAMBLE**

The purpose of this Agreement is to promote and maintain a harmonious relationship between the Employer and all employees covered by this Agreement with mutual respect between the Employer and the Union, to set forth terms and conditions of employment for all such employees with an amicable method of settling grievances or differences which may from time to time arise and to provide for the carrying on of the business under methods which will further the well being of employees (including their physical safety and psychological well being), together with efficiency and quality of operation, in order to ensure for the community a high standard of emergency health care.

### **ARTICLE 1 - DEFINITIONS**

# 1.01 <u>Definitions</u>

For the purpose of this Agreement:

- (a) "Bargaining Unit" means all employees within the scope of the bargaining unit as described in Article 3.01.
- (b) "Casual" means a person employed (I) on a call in basis or (2) on a relief basis to cover leaves up to three (3) months.
- (c) "EHS" means Emergency Health Services (NS).
- (d) "Employee" means a person who is included in the bargaining unit as defined in Article 3.01 and includes:
  - (i) "Full-time employee" is an employee who is hired for and regularly scheduled to work the full-time hours of work as provided in this Agreement on an indefinite basis.
  - (ii) "Regular part-time employee" is an employee who is hired for and regularly scheduled to work less than the full-time hours of work as provided in this Agreement on an indefinite basis.
    - (Regular part-time employees shall be entitled to benefits pro-rated on the basis of regularly scheduled hours of work, except as otherwise agreed to by the parties. For the purposes of earning entitlement to a benefit (for example, vacation, merit increments or pregnancy leave), calendar time of employment will be applicable.)
  - (iii) "Term employee" means a person who is hired for and regularly scheduled to work for a term in excess of three (3) months but not more than twenty-four (24) months.
- (e) "Employer" means EMC Emergency Medical Care Incorporated.

- (f) "Lockout" includes the closing of a place of employment, suspension of work or a refusal by the Employer to continue to employ a number of its employees done to compel its employees, or to aid another employer to compel its employees, to agree to terms or conditions of employment.
- (g) "Predecessor Employer" means an Ambulance Service which has been acquired by EMC.
- (h) "Site" means the base at which an employee is stationed.
- (i) "Spouse" means husband, wife, and common-law spouse. Common-law spouse includes a same sex partner in a common-law relationship except for the purpose of a pension plan where the pension plan contemplates otherwise.
- (j) "Strike" includes a cessation of work, or refusal to work or continue to work by employees in combination or in concert or in accordance with a common understanding, for the purpose of compelling their Employer to agree to terms or conditions of employment or to aid other employees in compelling their employer to agree to terms or conditions of employment.
- (k) "Supervisor" means a person appointed by the Employer as a supervisor and excluded in accordance with Subsection 2 of Section 2 of the *Trade Union Act*.
- (1) "Union" means the International Union of Operating Engineers, Local 968B.

### 1.02 Gender

Where the masculine gender is used it shall be read as including the feminine gender.

#### ARTICLE 2 - MANAGEMENT RIGHTS

#### 2.01 Management Rights

The Union recognizes and acknowledges that the operation and management of the service and direction of the work force are vested exclusively in the Employer unless specifically abridged, deleted or modified by this Agreement. Without restricting the generality of the foregoing, the Union acknowledges that it is the exclusive function of the Employer to:

- (a) maintain order and efficiency;
- (b) hire, promote, demote, classify, transfer, suspend or otherwise discipline employees for just cause provided that a claim by an employee that he/she has been disciplined without just cause may be the subject of a grievance and dealt with subject to the provisions of this Collective Agreement; and
- (c) make, enforce and alter, from time to time, reasonable rules and regulations to be observed by the employees.

#### 2.02 Consistent Application

The Employer shall not act in a manner inconsistent with the express provisions of this Agreement. The Employer shall exercise its rights in a reasonable manner.

### 2.03 Identification of Supervisory Officials

The Employer shall provide the Union with the names of its Supervisors, together with the geographic area for which each Supervisor is normally responsible so the Union is aware of the Supervisors with whom it should normally be dealing with in different parts of the Province.

### **ARTICLE 3 - RECOGNITION**

#### 3.01 Bargaining Unit Recognition

The Employer recognizes the Union as the exclusive bargaining agent for all full-time, regular part-time and term Paramedics in the bargaining unit as described in Certification Order L.R.B. 4619 or such further classification(s) as may be agreed upon, save and except:

- (a) Supervisors;
- (b) any other persons excluded by subsection (2) of Section 2 of the *Trade Union Act* of Nova Scotia;
- (c) casuals.

### 3.02 Mutual Agreements

No employee shall be required or permitted to make a written or verbal agreement with the Employer, or its representatives, which may conflict with the terms of this Collective Agreement.

### 3.03 Bargaining Unit Work - Supervisors

Supervisors (or other management personnel who are paramedics) shall not work on ambulances, except as **follows:** 

- (a) In recognition of the benefit of having Supervisors (and certain other management personnel who are paramedics) maintain their certification as paramedics and their skill level, Supervisors (and those other management personnel who are paramedics) may work on ambulances as follows:
  - (i) in Halifax-Dartmouth-Bedford-Sackville and Cape Breton Regional Municipality, Supervisors (or other management personnel who are paramedics) may work on a day ambulance on Friday or any other mutually agreeable day, with one (1) week's notice to a Union Business Representative;

SMSS\\1051363.1S

JO

- (ii) for all other areas, Supervisors (or other management personnel who are paramedics) may work on an ambulance for any open shift (for example, when an employee is on vacation or other scheduled day off), provided however:
  - (A) a Supervisor shall not bump any employee who has already been assigned the open shift;
  - (B) for any individual Supervisor, there shall be a maximum of twelve (12) shifts per contract year; and
  - (C) one (1) week's notice is to be provided to a Union Business Representative; and
- (iii) in any location, a Supervisor (or other management personnel who are paramedics) may work on an ambulance where an extra ambulance that is not normally scheduled is being used.
- (b) Nothing above restricts a Supervisor (or any other management personnel who are paramedics) from working on an ambulance:
  - (i) after the call-in list has been exhausted at straight time; or
  - (ii) in an emergency situation of wide scope such as a mass casualty incident.

### ARTICLE 4 - APPLICATION

4.01 This Agreement, including the Memoranda of Agreement and the Schedules which are attached, applies to and is binding on the Union, the employees and the Employer.

### **ARTICLE 5 - FUTURE LEGISLATION**

5.01 In the event that any law passed by the Legislature applying to the employees covered by this Agreement renders null and void any provision of this Agreement, the remaining provisions of the Agreement shall remain in effect for the term of the Agreement.

### ARTICLE 6 - NO DISCRIMINATION

6.01 No Discrimination for Union Activity

The parties agree that there will be no discrimination, interference, restriction, or coercion exercised or practiced with respect to any employee for reason of membership or legal activity in the Union.

### 6.02 No Discrimination - Human Rights

The parties agree that there shall be no discrimination against any employee on the basis of race, religion, creed, color, ethnic, national or aboriginal origin, sex, sexual orientation, source of income, political belief, affiliation or activity, family status, marital status, age, or physical disability or mental disability, contrary to the *Human Rights Act* (Nova Scotia).

### 6.03 Sexual and Personal Harassment

The Employer shall make all reasonable efforts to provide, and the Union and employees shall support, a workplace free from sexual harassment and any other harassment or abuse based on the protected characteristics set out in Article 6.02.

### ARTICLE 7 - RIGHTS AND PROHIBITIONS

#### 7.01 No Lockout or Strike

The Employer shall not cause a lockout and an employee shall not strike during the term of this Agreement.

### 7.02 No Sanction of Strike

The Union shall not sanction, encourage, or support financially or otherwise, a strike by its members or any of them who are governed by the provisions of this Agreement during the term of this Agreement.

# 7.03 Picket Line or Other Strike Activity

Notwithstanding any picket line or other strike activity, legal or illegal, by employees of any other employer, employees shall continue to perform their normal work duties.

### **ARTICLE 8 - CHECK OFF**

### 8.01 <u>Deduction</u>

The Employer agrees that, commencing the first pay in the calendar month following completion of the first month of employment, it will deduct from the bi-weekly earnings of all employees an amount equal to the regular dues payable by all members of the Union, as well as any special assessments uniformly required to be paid by all members of the Union.

### 8.02 Notification of Deduction

The Union shall inform the Employer in writing of the authorized deduction to be checked off for employees.

# 8.03 Remittance of Union Dues

- (a) Subject to paragraph (b), the amount deducted shall be forwarded to the Office Administrator of the Union no later than the month following the month in which deductions were made and shall be accompanied by a list showing all employees from whom deductions were made. At this time, the Employer shall also advise the Union in writing of all employees who have been hired or terminated the previous month.
- (b) Deductions for regular dues and any special assessments shall not apply to any employee who, because of his/her religious conviction or belief, cannot pay union dues and assessments, provided he/she makes a contribution equal to such union dues and assessments to some recognized charitable cause.

### 8.04 <u>Indemnification</u>

In consideration of the Employer deducting and remitting amounts to the Union, the Union agrees to indemnify and save the Employer harmless against any claim or liability arising out of or resulting from the operation of this Article.

#### 8.05 Income Tax Receipt for Dues

The Employer shall include on all **T-4** slips the amount deducted from employees' pay under this Article in the previous calendar year.

### ARTICLE 9 - UNION REPRESENTATION AND AFFAIRS

#### 9.01 Appointment of Stewards

The Employer acknowledges the right of the Union to appoint a reasonable number of Stewards. The Union's Business Representative shall advise the Employer's Director of Human Resources in writing of the names of Stewards, from time to time appointed, with a copy to the Employer's Regional Managers. The Employer shall not be required to recognize new Stewards until there has been five (5) business days' advance notification.

# 9.02 Steward's Duties

- (a) The Union shall designate one (1) Steward or alternate designate to deal with a Supervisor with respect to any complaint arising out of this Agreement or any general working conditions to be discussed with the Supervisor, subject to maintaining as much continuity as is reasonably possible.
- (b) A Steward may attend to his/her duties during working hours without loss of pay provided that:
  - (i) the duties do not disrupt the Steward's normal duties as an employee or the normal duties of any other employees;

- (ii) in order to ensure compliance with (i), no Steward shall leave his/her work site without obtaining the permission of his/her Supervisor or designate, such permission not to be unreasonably withheld; and
- (iii) the time shall be devoted to the prompt handling of necessary Union business.

#### 9.03 Union Officials

The Union's Business Representative or other Union official may enter a site office for the purpose of meeting with management or for the purpose of conferring with a Steward or an employee with respect to any dispute arising out of this Agreement provided:

- he obtains the prior approval of the Employer's Regional Manager or his/her representative, such approval not to be unreasonably withheld; and
- (b) such visits shall not interfere with normal operations.

## 9.04 Negotiating Committee

The Employer shall grant special leave for not more than six (6) representatives of the bargaining unit for the purpose of attending contract negotiation meetings with the Employer on behalf of the Union, where operational requirements permit, and on reasonable notice, such permission not to be unreasonably withheld. The Employer shall pay fifty percent (50%) of regular time lost from work for each of the above representatives to attend the meetings.

### ARTICLE 10 - EMPLOYEE INFORMATION

#### 10.01 Acquainting New Employees

The Employer agrees to acquaint new employees with the fact that the Collective Agreement **is** in effect and with the conditions of employment in Article 8 (Check Off). The Employer shall provide a copy of the Collective Agreement to each new employee.

## 10.02 Printing of Collective Agreement

Within a reasonable time of signing this Agreement, the Employer agrees to arrange for printing a copy of the Collective Agreement in booklet form and to supply a copy to all employees. The Union and the Employer shall share the cost of printing the Collective Agreement on a fifty-fifty (50/50) basis.

### 10.03 Bulletin Boards

The Employer shall allow the Union use of bulletin boards at each site for posting of notices of Union meetings and other Union activities. Such notices shall be dated and signed on behalf of the Union by the Business Manager or designate and shall normally remain posted for fourteen (14) calendar days. The Employer reserves the right to remove such notices, but should the Employer remove the notice prior to the expiry of the

fourteen (14) calendar days, an Employer's representative shall give notice to the site Steward (or in the absence of a Steward, to the Union), along with his/her reasoning for the removal.

### 10,04 EHS Provincial Medical Policies and Procedures Manual

- (a) The Employer will provide a copy of the EHS Provincial Medical Policies and Procedure Manual, with updates from time to time, for each ambulance.
- (b) The Employer will continue to provide a handbook (a "field guide edition" of the Manual) to each employee.

### 10.05 Job Descriptions

- (a) Upon request by an employee, the Employer shall provide to the employee the position description for the position held by the employee, provided however, the Employer is obliged to provide a position description to an employee only once during the term of this Agreement, unless the position description is changed, in which case the Employer is again obliged to provide the new position description to an employee once during the term of this Agreement.
- (b) The Employer shall ensure the position descriptions are reviewed, and revised if necessary, within one (1) year of the signing of this Agreement.
- (c) All revised position descriptions shall be provided to the Union within fifteen (15) days of any revision.
- (d) Reference to position descriptions in this Collective Agreement does not in any way restrict management rights with respect to job duties of a position or the assignment of work.

## ARTICLE 11 - GRIEVANCE PROCEDURE

### 11.01 Complaints

Any complaint by an employee as to the interpretation or application of the provisions of this Agreement shall be considered a grievance, provided that:

- (a) no complaint by an employee shall be considered a grievance until the employee has discussed his/her complaint with his/her immediate supervisor and the immediate supervisor has had twenty-four (24) hours to adjust the complaint with the employee, with or without a Steward, (provided however this requirement for a discussion does not apply in the case of discipline which has been imposed), and
- (b) the complaint is submitted in writing in accordance with Article 11.02.

# 11.02 Steps in Grievance Procedure

The following procedure shall be adhered to in processing grievances: SMSSW1051363.18

Step No. 1 The employee and/or Steward shall present the grievance in writing to the employee's supervisor or designate within twelve (12) days, including any time required under Article 11.01 (a), of the occurrence of the event giving rise to the grievance, or of the time the employee could reasonably be expected to be aware of the occurrence of the event giving rise to the grievance, but in no case more than thirty (30) days after the occurrence of the event. The grievance shall be on a form approved by the Employer and the Union and shall contain the name of the grievor, the date upon which the grievance was prepared, the nature of the grievance and the date(s) upon which it is alleged to have occurred or arisen, the article and subsections of the Collective Agreement which the grievor alleges to have been violated, the remedy sought and the signature of the grievor, together with the signature of either the Union Steward or the Union's Business Representative or the designated representative of either; (there are two (2) mandatory signatures, one being the grievor and the second being the Union representative). The Supervisor shall reply giving his/her decision in writing within five (5) days following receipt of the grievance.

Step No. 2 If the grievance is not satisfactorily resolved at Step No. 1, the Union's Business Representative or designated representative shall present the grievance in writing in accordance with the form set out in Step No. 1 to the Employer's Director of Human Resources or her designated representative. The grievance shall be presented within the earlier of five (5) days of receipt of the reply at Step No. I or the deadline for such reply if not submitted by the Employer. The Director of Human Resources and the Union's Business Representative, or their designated representatives, shall make all reasonable efforts to meet within ten (10) days after presentation of the grievance at Step No. 2, to discuss the grievance. The meeting shall take place at premises of the Employer or such other facilities as may be arranged by the Employer. The Director of Human Resources or her designated representative shall reply giving her decision in writing within ten (10) days following a meeting, or if a meeting cannot be arranged, within ten (10) days of the expiry period for the meeting.

Step No. 3 If the grievance is not satisfactorily resolved at Step No. 2, the Union may, within the earlier of twenty (20) days of the reply at Step No. 2 or the deadline for such a reply should no such reply be submitted by the Employer, refer the grievance to arbitration.

#### 11.03 Suspension or Discharge Grievances

A grievance alleging suspension or discharge without just cause shall be submitted at Step No. 2 of the grievance procedure. The employee shall present this grievance in the form set out in Step No. 1 above within five (5) days from the date of the suspension or discharge.

# 11.04 Union Policy Grievance

A Union policy grievance, defined as a dispute with respect to the general application,  $\frac{1}{2}$ 

interpretation or administration of this Agreement, may be lodged by the Union's Business Representative or designate in writing with the Director of Human Resources at Step No. 2 of the grievance procedure at any time within twenty (20) days of the occurrence of the circumstances giving rise to the grievance; and if it is not satisfactorily settled it may be processed to arbitration in the same manner and to the same extent as the grievance of an employee. This paragraph does not apply in cases of individual grievances and may not be used to by-pass the grievance procedure set out in the foregoing paragraphs of this Article.

### 11.05 Employer Grievance

Any complaint of the Employer with respect to the general application, interpretation or administration of this Agreement shall also be considered a grievance to be dealt with in the same manner as an employee grievance, except that the Director of Human Resources of the Employer or her designated representative shall first discuss the complaint with the Union's Business Representative and failing an answer or satisfactory settlement, the Director of Human Resources of the Employer or her designated representative may submit the grievance in writing to the Union's Business Representative.

#### 11.06 Time Limits

- (a) The time limits specified in this Article shall be deemed to be exclusive of Saturdays, Sunday and holidays.
- (b) Any of the time limits provided in this Article may be extended by mutual agreement in writing between the Employer and the Union.
- (c) The time limit for the initial submission of a grievance, whether it be at Step No. 1 or Step No. 2, is mandatory (that is, failure by the grieving party to comply with the time limit shall result in the grievance being deemed abandoned and terminated with all rights of recourse to the grievance procedure at an end). Subsequent time limits are directory (that is, the arbitration board or single arbitrator shall be able to overrule a preliminary objection that the time limit was missed, providing that the board or arbitrator is satisfied that the grievance has been handled with reasonable dispatch and the position of either party is not significantly prejudiced by the delay).

### 11.07 Grievance Hearings

In any hearing at any step of the grievance procedure or arbitration procedure, the grievor shall have the right to be present and shall suffer no loss of regular pay.

### ARTICLE 12 - ARBITRATION PROCEDURE

### 12.01 Reference to Arbitration

Either of the parties may, after exhausting the grievance procedure in Article 11, notify the other party within twenty (20) days of the receipt of the reply at Step No. 2 or the

SMSS\105136315

deadline for receipt of such reply, of its desire to refer the grievance to arbitration pursuant to provisions of the *Trade Union Act* (Nova Scotia) and this Agreement.

#### 12.02 Single Arbitrator Unless Board Requested

In the event that a grievance is submitted to arbitration, it shall be heard by a single arbitrator, unless either party requests that it be heard by a three member arbitration board

### 12.03 Single Arbitrator

If a grievance is to be heard by a single arbitrator and the Union and the Employer fail to agree upon the appointment of an arbitrator within seven (7) days of notice of arbitration being given in accordance with Article 12.01, either party may apply to the Minister of Labour for Nova Scotia for appointment of the arbitrator in accordance with the provisions in the *Trade UnionAct* (Nova Scotia).

### 12.04 Arbitration Board

If either party requests that a grievance be heard by a three member arbitration board, the Union and the Employer shall each appoint a member of the arbitration board within seven (7) days of notice of arbitration being given in accordance with Article 12.01. Should either party fail to appoint a nominee or if the two nominees fail to agree upon a chair within ten (10) days of the appointment of the second of them, then either party may apply to the Minister of Labour for Nova Scotia for appointment of the chair or such other appointment as is necessary to constitute the board of arbitration.

### 12.05 Authority of Arbitrator/Board

The single arbitrator or board of arbitration, as the case may be, shall not have any authority to alter, modify or amend any part of this Agreement, or to give any decision contrary to the terms or conditions of this Agreement.

### 12.06 Hearing and Decision

The single arbitrator or board of arbitration, as the case may be, shall hear and determine the difference or allegation, including any question as to whether the difference is arbitrable, and shall issue a decision. All decisions arrived at by a single arbitrator or board shall be final and binding upon the Employer, the Union and the employee or employees concerned. The decision of the majority is the decision of the board.

### 12.07 Fees and Disbursements of Arbitrator/Board

Each of the parties to the Agreement will pay the fees and disbursements of its nominee to the board of arbitration, and will share equally the fees and disbursements of the chair or single arbitrator.

### 12.08 Time Limits

- (a) The time limits specified in this Article shall be deemed to be exclusive of Saturday, Sunday and holidays.
- (b) Any of the time limits provided in this Article may be extended by mutual agreement in writing between the Employer and the Union.

### **ARTICLE 13 - RESIGNATION**

### 13.01 Notice of Resignation

An employee will provide the Employer with two (2) weeks' notification in writing of their intention to resign from their employment with EMC, provided however the Employer may accept a shorter period of notice.

### 13.02 Failure to Give Notice

- (a) An employee who fails to give notice required by Article 13.01 shall be struck from the payroll effective the date he/she absents himself without leave, and shall have deducted from monies owed him by the Employer from all sources, including any vacation pay, a sum equivalent to the salary payable to him for the period of notice which he/she failed to work.
- (b) If the employee is reinstated in accordance with 13.03, then any deductions made pursuant to 13.02 (a) shall be reinstated.

### 13.03 Withdrawal of Resignation

An employee who has resigned as a result of anxiety or similar emotional state after a traumatic event arising in the course of employment may withdraw his/her resignation within seventy-two (72) hours from the time it was submitted to the Employer.

# ARTICLE 14 - PERFORMANCE APPRAISALS

- 14.01 (a) The Performance Appraisal System provides a framework for ongoing communication between employees and their supervisors regarding the employee's job performance and satisfaction. Performance appraisals will be used for employee development purposes only. Performance appraisals shall not be used by the Employer to adversely affect the employees' promotional opportunities or as the basis of discipline.
  - (b) Performance appraisals will be conducted on a regular predetermined schedule and will be conducted during the regular working hours of the employee.
  - (c) The performance appraisal interview is a two-way dialogue that will include an opportunity for the employee to provide feedback to their supervisor on a voluntary basis. An employee shall be given at least twenty-four (24) hours'

SMSS\\105136315

- notice of the performance appraisal meeting with such meeting to take place during an employee's regular hours of work, unless the employee agrees to less notice and/or a different time.
- (d) The criteria used to evaluate an employee's performance in the performance appraisal must reflect the job the employee performs as it relates to the goals and values of the Employer in the delivery of patient care.
- (e) An employee who chooses not to accept extra workload and/or duties (e.g., extra shifts or volunteer positions including interim SOP), will not, as a result, receive an unacceptable evaluation. However, an employee who chooses to accept extra workload and/or duties may, as a result, be commended and the extra workload and/or duties may have a positive affect on that employee's evaluation.
- (f) Where an employee indicates satisfaction with their current position and does not express an interest in future promotion or advancement, these statements or views shall not be construed against the employee in any respect. Such opinions or views of the employee may be subject to change at a future date. An employee's comments on their performance appraisal shall not be the sole factor considered in determining which employees shall be given training opportunities.
- (g) The employee shall be given a copy of any performance appraisal before it is placed in their file. The employee shall have the right to respond in writing to the performance appraisal, and such response shall form part of the employee's file.

### ARTICLE 15 - TIME OFF FOR UNION BUSINESS

### 15.01 <u>Leave for Union Business</u>

- (a) Employees who are elected to the following positions may be granted leaves of absences without pay in accordance with the conditions set out in paragraph (c):
  - as members of the Executive Board of the Union for the attendance at Board meetings;
  - (ii) as delegates to attend conventions of the Union's affiliated bodies including, IUOE, CLC, Nova Scotia Federation of Labour;
  - (iii) as members of standing Committees of the Union for the attendance at meetings of standing Committees;
  - (iv) as members of the Executive to attend Executive Meetings of the Nova Scotia Federation of Labour.
- (b) Employees who are selected or appointed to attend Union educational programs or to work as replacements for Union staff on a relief basis and for such other proposed reasons as may be agreed to by the Employer may be granted a leave without pay in accordance with the conditions set out in paragraph (c).

- (c) The leaves of absence in paragraphs (a) and (b) are subject to the following conditions:
  - (i) the Employer receives reasonable notice;
  - (ii) operational requirements; and
  - (iii) permission being granted by the Employer, permission not to be unreasonably refused.

# 15.02 Notification To Employer

The Union shall notify the Employer of the names, including the site wherein the employee **is** employed, of the members of the Executive Board and Bargaining Committee.

### 15.03 Annual Meeting

- (a) Where operational requirements permit and on reasonable notice, the Employer shall grant special leave without pay for a period not exceeding two (2) days, and special leave without pay for traveling time for such portion of the working day prior to and following the meeting as may be required to employees who are elected or appointed as registered delegates to attend the Annual Meeting of the Union. Such permission shall not be unreasonably withheld.
- (b) The Union shall notify the Employer, in writing, of the names, including the site wherein the employee is employed, of the registered delegates to the Annual Meeting of the Union at least three (3) weeks in advance.

### 15.04 Salary Continuance

The Employer will continue the salary of an employee who is granted leave without pay in accordance with Article 15 and will bill the Union for the employee's salary.

# 15.05 Arbitrations and Joint Consultation

- (a) Where operational requirements permit, and on reasonable notice, the Employer shall grant special leave with pay to employees who are called as a witness by an arbitrator or an arbitration board as prescribed by Article 12. The Union will reimburse the Employer for one-half of the salary.
- (b) Where operational requirements permit, and on reasonable notice, the Employer shall grant special leave with pay to employees who attend meetings with management in a joint consultation as prescribed in Article 22.

### 15.06 Leave of Absence for Union President

An extended leave of absence for the President of the Union shall be granted in accordance with the following:

- (a) An employee who declares his/her intention to offer for the position of President of the Union shall notify the Employer as soon as possible after declaring his/her intention to seek the office of the President.
- (b) An employee elected or appointed as President of the Union may, on request of the Union's Business Manager, be given leave of absence without pay.
- (c) For the purposes of paragraph (b), the leave of absence
  - (i) shall be for six (6) months or longer;
  - (ii) shall commence and end as determined by the Union, provided one (1) month's notice of both the commencement and expiry of the leave is provided to the Employer.
- (d) All employee benefits (Article 31) of the employee shall continue in effect while the employee is on leave of absence and the Union shall reimburse the Employer for the Employer's share of the contributions.
- (e) Notwithstanding paragraphs (b) and (e), the gross salary of the President shall be determined by the Union and paid to the President by the Employer, and the amount of this gross salary shall be reimbursed to the Employer by the Union. The Union shall also reimburse to the Employer the Employer's share of contributions for El premiums, Canada Pension Plan, other pension and group insurance premiums made on behalf of the employee during the period of leave of absence.
- (f) Upon expiration of his/her leave of absence, the employee shall be reinstated in the position he/she held immediately prior to the commencement of leave, or if the position no longer exists, to another position in accordance with this Agreement.
- (g) Notwithstanding paragraph (b) or any provision of this Agreement to the contrary, the period of leave of absence shall be deemed to be continuous service with the Employer for all purposes.
- (h) Notwithstanding the provisions of the Agreement, vacation earned but not used prior to taking office shall be carried over to be taken in the fiscal year in which the employee returns from leave of absence.

#### 15.07 No Loss of Service. Seniority or Benefits

While on leave for Union business pursuant to this Article, an employee shall continue to accrue and accumulate service and seniority credits for the duration of the leave, and the employee's service and seniority shall be deemed to be continuous. There shall be no loss of benefits while on Union business pursuant to this Article.

### **ARTICLE 16 - APPOINTMENT**

#### 16.01 Probation

- (a) New employees shall serve a probationary period as follows:
  - (i) Full-time employee: six (6) months; and
  - (ii) Part-time employee: number of hours equivalent to hours worked by a full-time employee in the employee's location, but no more than twelve (12) months.
- (b) The Employer may extend an employee's probationary period for three (3) months.
- (c) A previous permanent employee (who had successfully completed his/her probationary period) whose employment was terminated for any reason and who is re-employed within twelve (12) months from the date of such termination shall not be required to undergo a second probationary period.

# 16.02 <u>Dismissal of Probationary Employee</u>

- (a) An employee can be dismissed during his/her probationary period, including any extension in accordance with 16.01 (b) above, if the Employer determines that the employee is unsatisfactory and such dismissal shall be deemed to be for just cause.
- (b) If an employee is dismissed during his/her probationary period, the Employer
  - (i) advise the employee of the reason(s) in general; and
  - (ii) where the dismissal is for reasons other than willful misconduct or disobedience or neglect of duty, provide the employee with one (1) week's notice and/or pay in lieu of notice.
- (c) The Employer shall notify the Union when a probationary employee is terminated.

# 16.03 <u>Union Notification</u>

The Employer shall notify the Union, in writing, of all appointments, lay-offs, recalls and terminations.

### ARTICLE 17 - DISCHARGE. SUSPENSION AND DISCIPLINE

# 17.01 Just Cause

No employee who has completed his/her probationary period shall be disciplined, SMSSN105136315

suspended without pay or discharged except for just and sufficient cause.

#### 17.02 Notification

Where an employee is disciplined, suspended without pay or discharged, the Employer shall, within five (5) business days of the discipline, suspension or discharge notify the employee and the Union in writing by registered mail, personal service or facsimile stating the reason for the discipline, suspension or discharge.

#### 17.03 Record of Disciplinary Action

- (a) The Employer agrees not to introduce as evidence in a hearing relating to disciplinary action, any document from the file of an employee, the existence of which the employee was not aware at the time of filing.
- (b) Notice of a disciplinary action which may have been placed on the human resources file of an employee shall be destroyed after three (3) years have elapsed since the disciplinary action was taken provided that no further disciplinary action has been recorded during this period.

### 17.04 Employee Access to Human Resources File

- (a) An employee, upon giving forty-eight (48) hours' notice, may examine their human resources file in their respective administrative areas and within seventytwo (72) hours in the Human Resource Department. For the purpose of this provision the notice period shall not include Saturdays, Sundays or Statutory Holidays. Complaints about an employee from outside the Employer's place of business which are placed in the employee's human resources file shall be shown or made available to the employee. The employee shall have the right to make copies of any material contained in his/her file not previously provided to the employee.
- (b) The human resources records of an employee shall not be shared in any manner with any other employer or agency, without the prior written consent of the employee concerned except for EHS (record related only to clinical issues, registration or driving record) or as required by law.

### 17.05 Right to Have Steward Present

- (a) When management meets with an employee to discipline the employee (other than in the cases of where spontaneous discipline may be warranted), the employee shall have a right to have a Steward present, provided the meeting is not unreasonably delayed to allow for the presence of the Steward.
- (b) Verbal warnings are not considered discipline and shall not form part of the Human Resources file. Employees shall not have the right to have a Steward present when issued a verbal warning.

SMSSW1051363.15

## **ARTICLE 18 - SENIORITY**

#### 18.01 Seniority Defined

Seniority shall be defined in accordance with the following:

- (a) Seniority shall mean the length of continuous employment in a bargaining unit position as an employee with the Employer including any Predecessor Employer (and in the event of loss of seniority, from the most recent date of hire as an employee).
- (b) <u>Former "non-permanent employees":</u>

The continuous employment of formerly "non-permanent employees" is considered to have been broken in a year in which they did not work or receive pay for at least sixty (60%) of full-time hours. For such employees, their date of hire, for the purpose of establishing seniority as of the time of transfer from a predecessor employer to the Employer, shall be their actual date of hire where there is no break in continuous employment or January 1st of the year following the most recent break in continuous employment.

### 18.02 Posting of Seniority Lists

- (a) Annually on February 1, the Employer shall post a list setting out each employee's seniority date. Each employee shall have thirty (30) days from the date the list is posted to challenge his/her seniority date in writing, The Employer shall reply to the employee's written objection within thirty (30) days of receipt of the written objection. If no written objection is received by the Employer within thirty (30) days from the date the list is posted, the seniority date on the list shall be the employee's seniority date for all purposes following the posting of the list.
- (b) An employee who is absent from work for two (2) or more calendar weeks of the thirty (30) day posting period shall have the posting period extended by the equivalent number of days from the date of his/her return to work to object in writing to his/her seniority date. However, until and unless such written objection is received by the Employer, and in any event no later than thirty (30) days from the employee's return to work, the posted seniority date for the employee will be considered to be the employee's seniority date for all purposes.
- (c) Once an employee unsuccessfully challenges his/her seniority date, the employee may not dispute his/her seniority date unless there is new material evidence which could not reasonably be expected to have been available at the earlier time.
- (d) The parties agree to the expedited process to resolve disputes regarding seniority as outlined in Memorandum of Agreement #1.

#### 18.03 Loss of Seniority

An employee shall no longer be considered an employee and his/her seniority shall cease

if he:

- resigns verbally or in writing and does not withdraw such resignation within seventy-two (72) hours, in accordance with Article 13.03;
- (b) is discharged and is not reinstated through the grievance or arbitration procedure;
- (c) is absent and does not notify the Supervisor or his/her delegate within four (4) consecutive working days, unless such notification is not possible;
- (d) is laid off for a period in excess of twelve (12) months;
- (e) fails to respond to a notice of recall within two (2) working days after receipt of the recall by registered mail or fails to report for work within fourteen (14) calendar days after being advised of his/her recall, unless he/she is unable to return due to illness or injury and so notifies the Supervisor. This provision shall have no application in the case where an employee refused recall for casual work or term employment of less than twelve (12) months if the employee is employed elsewhere; or
- (f) is absent from work as a result of illness or accident for a period in excess of thirty-six (36) months; or returns from disability to the Recall List after hisher first twenty-four (24) months of disability, but prior to the expiry of thirty-six (36) months from his/her initial date of disability, and his/her recall rights expire in accordance with Article 21.12; (it is understood that after twenty-four (24) months of disability, but prior to the expiry of thirty-six (36) months, an employee is not entitled to return to hisher regular site, but is only entitled to be placed on the Recall List pursuant to Article 21.11).

#### 18.04 Service

For the purpose of this Agreement, "service" means:

- (a) (i) the service with which an employee was credited with as an employee with a Predecessor Employer at the time of transfer to the Employer; and
  - (ii) total accumulated months of employment with the Employer; and
  - (iii) other service credits which may be recognized by the Employer (not predecessor Employer) on hiring of an employee such as credits recognized for time worked with other Employers as a paramedic.
- (b) Service credits shall be prorated for regular part-time employees and employees who work part-time during a term appointment.

### 18.05 Service Credits

Service credits shall be used for the purpose of determining service related benefits such as vacation and severance pay.

# **ARTICLE 19 - HOLIDAYS**

### 19.01 Holidays

(a) The Employer agrees that the following shall be designated as paid holidays:

New Year's Day
Good Friday
Easter Sunday
Victoria Day
Canada Day

Labour Day
Thanksgiving Day
Remembrance Day
Christmas Day
Boxing Day

One (1) additional day in each year that, in the opinion of the Employer, is recognized to be a provincial or civic holiday in the area in which the employee is employed, or, in any area where, in the opinion of the Employer, no such additional day is recognized as a provincial or civic holiday, the first Monday in August.

Any other day declared by the Province of Nova Scotia to be a general holiday.

- (b) Holiday means in the case of a shift that does not commence and end in the same day, the twenty-four (24) hour period commencing at the start of the day shift on the date of holiday and ending twenty-four (24) hours later.
- (c) It is understood that an employee who is not scheduled to work on the holiday shall be paid for the entire shift an employee would normally work.

### 19.02 Exception

Article 19.01 does not apply to an employee who is absent without pay on either his/her working day immediately preceding or his/her working day immediately following the designated holiday (for example, employees on unpaid leave, suspension without pay or layoff are not eligible).

### 19.03 Holiday Coinciding with Paid Leave

Where a day that is a designated holiday for an employee as defined in Article 19.01, falls within a period of leave with pay, the holiday shall not count as a day of leave.

#### 19.04 Compensation for Work on a Holiday

Where an employee is regularly scheduled to work in accordance with Article 27 and his/her regularly scheduled day of work falls on a paid holiday, as defined in Article 19.01, or he/she is called back to work on a paid holiday, he/she shall receive compensation equal to two and one-half (2%) times his/her regular rate of pay as follows:

(a) (i) compensation at one and one-half (1½) times his/her regular rate of pay, for the hours worked on the holiday; plus

- (ii) pay which the Employee would have received if he/she had not worked on a holiday.
- (b) Notwithstanding (a)(ii) above, an employee, upon his/her request, may take time off with pay in lieu of the holiday based on regular hours of work with such time off to be banked and available for payout in accordance with Article 28,06.

### 19.05 Christmas Day or New Year's Off

- (a) Employees who are scheduled to work on both Christmas Day and New Year's Day may, prior to November 30, request to be scheduled off for one of the days.
- (b) Employees who are scheduled to work Christmas Day after having worked Christmas Day the previous year may, prior to November 30, request to be scheduled off on Christmas Day.
- (c) Upon receipt of a request under paragraph (a) or (b), the Employer will make reasonable efforts, subject to operational requirements, to accommodate the request by scheduling a replacement.

### **ARTICLE 20 - JOB POSTING**

### 20.01 Job Posting

- (a) When a new position, permanent vacancy or term position is created within the bargaining unit, the Employer shall post a notice on bulletin boards for a minimum of two (2) weeks. If the Employer becomes aware that there will be a permanent vacancy arising within four (4) weeks (for example, an employee has submitted a written resignation effective a certain date), the Employer will post the notice at the time of the next weekly postings.
- (b) The notice posted shall include:
  - (i) the classification and work area;
  - (ii) whether the posting is for a permanent position or a term position;
  - (iii) in the case of a term position, the expected duration;
  - (iv) whether the appointment is full-time, part-time and any applicable part-time designation; and
  - (v) the hourly rate of pay.
- (c) Only those postings which cannot be filled with a qualified employee from the bargaining unit will be available for filling from outside the bargaining unit.

### 20.02 Job Postings (General)

- (a) When a permanent job vacancy occurs, the Employer shall post the job -Paramedics by certification level (PCP, ICP, ACP) or Senior Operations Paramedic - for fourteen (14) calendar days. An employee desiring the position must make application in writing to management within fourteen (14) days of the first day of posting.
- (b) (i) Job postings for "PCP" are for PCP applicants only, not for paramedics in a higher classification.
  - (ii) Job postings for "PCP or higher" may be awarded to a paramedic in a classification higher than PCP only if there are no suitable PCP applicants, in which case preference is given first to ICP applicants and then to ACP applicants.
  - (iii) Job postings for "ACP or lower" may be awarded to a paramedic in a classification lower than ACP only if there are no suitable ACP applicants, in which case preference is then given first to ICP applicants and then to PCP applicants.
  - (iv) Where a job posting is classification specific and there are no suitable employee applicants in the classification, the Employer may at its option repost the position or may hire from outside the bargaining unit.
- (c) The Employer may postpone the start time for the successful applicant in the posted position for operational reasons such as difficulty in obtaining the ALS replacement for up to three (3) months.
- (d) (i) If there is a permanent job vacancy at a site (and therefore a regularly scheduled shift which is open in that no individual employee is regularly assigned to the shift), a Full-time employee already working within the site may, within the posting period for the vacancy, request in writing to be regularly assigned to the open shift.
  - (ii) The Employer shall consider such request, taking into account any factors which the Employer considers relevant, including classification, seniority, experience as a paramedic and knowledge of the site. If the Employer denies the request, the Employer shall advise the employee in writing of the reason(s) for the denial, with a copy to the Union's Business Manager.

# 20.03 Filling Position for a Field Paramedic

Selection of persons to fill a posted job which may involve a lateral transfer will be governed by the following conditions:

(a) Until an employee (including a term employee) has completed his/her probationary period, the employee is not eligible to apply for a posting for another site unless the other site has the same Supervisor.

- (b) Selection of persons to fill the posted job shall be based on seniority among qualified applicants (PCP, ICP or ACP as the case may be), except that:
  - (i) an employee who has completed his/her probationary period but has not completed 12 months of full-time employment (or equivalent hours for a Part-Time employee) at a site may apply for a posting at another site, but will not be considered for selection to fill the posting unless the posting is not filled by employees who have met this eligibility requirement; and
  - (ii) ICPs and ACPs who have completed less than twenty-four (24) months of employment as a Full-Time employee (or equivalent hours of part-time employment) may be considered, but may be refused a posted position for operational reasons such as difficulty in obtaining an ALS replacement at the site where the employee is working.
- (c) Term employees are subject to the following conditions:
  - (i) Term employees who have not completed twelve (12) months of their term can only apply for a permanent position when they have three (3) months or less left in their term and may be required to complete their term before moving to the permanent position, unless the permanent position is in the same site; and
  - (ii) Term employees can only apply for another term position during the last thirty (30) days of their term and may be required to complete their term before moving to a new term position unless the new term position is at the same site.

On expiry of a term position, the former Term employee shall, for a period of ninety (90) calendar days, or such earlier time as the former Term employee accepts another position with the Employer, be given preference over a Casual employee for any posted position for which the former Term employee applies. It is understood that such preference does not give the individual status as an employee and, on receiving a posted position, a former Term employee is subject to a new probationary period unless he/she has successfully completed his/her probationary period and has received a posted position within the ninety (90) calendar days.

### 20.04 Promotion

In selecting persons to fill a posted job which involves a promotion (for example, Senior Operations Paramedics such as Medical Operations Paramedic, Fleet Paramedic, Procurement Paramedic, etc.) the following conditions apply:

- selection shall be based on merit (including skill and ability) and where applicants are of equal merit, seniority shall govern; and
- (b) preference shall be given to applicants from within the Administrative Area as set out in Schedule "C", but the Employer may fill the vacancy from outside the

Administrative Area where there are no suitable applicants from within the Administrative Area.

#### 20.05 Temporary Transfers

- (a) In circumstances where there is a need to temporarily transfer employees to any part of the Province for a period not to exceed sixty (60) calendar days the following shall apply:
  - (i) The Employer shall notify employees by inviting expressions of interest.
  - (ii) Preference for the transfer will be given first to the most senior qualified employee as per Article 28.08 (a) within the Region who expressed an interest in the temporary transfer, provided operational requirements permit.
  - (iii) Any employee who accepts a temporary transfer in accordance with this Article shall receive fifteen percent (15%) over and above his/her hourly rate for the length of the temporary transfer. In addition, an employee shall receive reasonable expenses incurred as a result of the temporary transfer which shall include meals and kilometrage, and other reasonable expenses which may include accommodations.
- (b) Where there is no response to the expression of interest, the most junior qualified employee as per Article 28.08(a) will be transferred by reverse seniority within the region provided operational requirements permit and there is no undue hardship placed on the employee as a result of the temporary transfer. The Employer shall not exercise the right to temporarily transfer an employee in an unreasonable or arbitrary manner.
- (c) Notwithstanding the above, if the circumstances are of an urgent nature or an emergency the Employer shall use the Call Back System in accordance with Article 28.08 until the completion of (a) and (b) above.
- (d) The Employer's right to transfer employees in accordance with this provision shall not be used to avoid the posting of positions in accordance with Article 20.01.

### 20.06 Reassignments

Employees shall report for work for regularly scheduled shifts at their regular site (which may vary in the case of employees who regularly rotate through various sites), unless mutually agreed otherwise, but it is understood that after reporting for work an employee may be assigned to work out of another site in which event the Employer will provide transportation to and from that other site or, if there is mutual agreement for the employee to utilize his/her own vehicle, pay kilometrage for travel to and from that other site.

#### 20.07 Secondments

- (a) Where it is determined by the Employer that bargaining unit employees will be considered for secondment opportunity, either with EMC or its sister company, Emergency Medical Services Inc., the Employer will notify the employees by posting an Expression of Interest (other than where an employee is assigned light duties on a temporary basis). Any bargaining unit employee who accepts the secondment opportunity shall be covered under all provisions of the Collective Agreement in addition to receiving any additional remuneration applicable to the position to which the employee is seconded. (It is understood that allowing bargaining unit employees secondment opportunities does not result in the work being done during the secondment being considered bargaining unit work.)
- (b) A secondment means a temporary assignment to a position outside the bargaining unit. The initial secondment will not exceed one (1) year, but may be extended for up to an additional year.
- (c) When an employee is seconded:
  - (i) If the secondment is for more than three (3) months and the secondment is for a full-time position (which, for the purposes of this Article, means forty (40) hours per week or more), then the Employer will post a term position.
  - (ii) If the secondment is for more than three (3) months and if the secondment is for a full-time position, then during the period of the secondment:
    - (A) the employee will be removed from his/her regular scheduled shifts at his/her regular site;
    - (B) the employee will be permitted to participate in the call-in system through the availability lists provided, however:
      - the seconded employee has provided his/her availability dates:
      - the employee is not eligible to work a call-in during the ten (10) hour period immediately before he/she is on duty in the seconded position.

Examples (for illustration purposes only):

Example I: An employee who is seconded to EMC's Dartmouth office for three (3) months and who is working Monday through Friday from 8 a.m. to 5 p.m. (a full-time position):

 will be taken out of their regular rotation in the field for the three (3) month period;

- is not eligible for call-ins for the period starting Monday until after he has completed his work relating to the secondment on Friday;
- is eligible for call-ins on weekends, subject to the ten (10) hour limitation.

Example 2: An employee seconded to EMS to work on an oil rig for six (6) months on a cycle of two (2) weeks on and two (2) weeks off (a full-time position):

- is removed from their regular rotation in the field for the full six (6) months;
- is not eligible for call-ins during the two (2) weeks when they are on duty;
- is eligible for call-ins during the two (2) weeks when they are ashore, subject to the ten (10) hour limitation period.
- (iii) If the secondment is for more than three (3) months, but not for a full-time position (example: an individual is seconded to EMC's Bedford office for three (3) days each week), the Company may post for a part-time term position.
- (iv) If the secondment is for less than three (3) months, either full-time or parttime, then the Company will attempt to fill the regular shifts of the seconded employee through the call-in system.
- (v) Where the secondment is not for a full-time position in excess of three (3) months, but is for more than one (1) month, the Employer will discuss with the Union's Business Representative whether the secondment creates any significant adverse effects on employees at the site where the seconded employee is normally based (for example, whether the secondment creates significant difficulties for employees to obtain leaves) and whether the Employer should take steps to address that adverse effect.

### 20.08 Trial Period

An employee who accepts a bargaining unit position as Senior Operations Paramedic as a result of a job posting shall be placed in the position for a trial period not exceeding three (3) months. If the employee proves unsatisfactory or wishes to return to his/her former position, he/she shall be returned to his/her former position. In the case of an employee who wishes to return to his/her former position, he/she shall advise the Employer in writing and shall be returned to his/her former position within two (2) weeks of the Employer receiving such notification, or such other time as mutually agreed. The former position shall be filled on a casual basis during the trial period.

### 20.09 Non-Bargaining Unit Vacant Position

An employee promoted to a position outside the bargaining unit, shall be placed in a position for a trial period of six (6) months. During the six (6) month period the Employer may elect to return such employee to the bargaining unit or the employee may elect to return to the bargaining unit. During the six (6) month period the employee shall continue to accumulate service and seniority credits for all purposes.

### 20.10 No Avoidance

The Employer will not use casual employees for the purpose of avoiding the filling of a vacancy in the bargaining unit.

### ARTICLE 21 - JOB SECURITY

### 21.01 Joint Committee on Labour Relations

- (a) Within sixty (60) days of the signing of this Agreement, the parties are to establish a Joint Committee on Labour Relations of equal representation of the Union and the Employer for the purpose of maintaining continuing cooperation and consultation on technological change and job security. The Union representation will include the Business Representative and a designated representative of the Union. The committee may appoint additional representatives as required.
- (b) The Joint Committee on Labour Relations shall consult as required to discuss matters of concern between the parties related to technological change and circumstances identified in Article 21.01(d).
- (c) The Joint Committee on Labour Relations shall be responsible for:
  - (1) defining problems;
  - (2) developing viable solutions to such problems; and
  - (3) recommending the proposed solution to the Employer.
- (d) The Employer will provide the Joint Committee on Labour Relations with as much notice as reasonably possible of expected redundancies, relocations, reorganizational plans, technological change and any other circumstances which may impact the job security of bargaining unit employees with a view toward minimizing the adverse effects on employees.
- (e) Time spent by employees traveling to and from meetings of the Joint Committee on Labour Relations and in attendance at these meetings shall be considered as time worked and compensated in accordance with the Collective Agreement.

### 21.02 Definition

For the purposes of this Article, "technological change" means the introduction of equipment or material by the Employer into its operations, which is likely to affect the job security of employees.

### 21.03 Introduction

The Employer agrees that it will endeavor to introduce technological change in a manner which, as much as is practicable, will minimize the disruptive effects on employees and services to the public.

### 21.04 Notice to Union

The Employer will give the Union written notice of technological change at **least** three (3) months prior to the date the change is to be effected. During this period the parties will meet to discuss the steps to be taken to assist employees whose job security could be affected.

### 21.05 Training

- (a) The Employer shall bulletin, at least thirty (30) days in advance, any training courses and experimental programs for employees. The bulletin shall contain the following information:
  - (i) type of course (general description of subject and materials to be covered);
  - (ii) time, estimated duration, and general location of the course;
  - (iii) basic minimum qualifications required for applicants.

The Employer reserves the right to cancel or postpone these courses and programs.

- (b) The Employer shall pay for all training programs required or mandated by the Employer and all time in attendance at such programs and practicums shall be considered time worked and compensated in accordance with the pay article of the Collective Agreement. The Employer also agrees to pay all related expenses incurred through participation in mandated training programs.
- (c) Employees who are laid off shall be offered the opportunity for training which would keep their credentials current. The Employer shall pay for the first day of training only.

### (d) <u>Clinical Education Assistance</u>

In April of each year, upon presentation of a current EHS registration card, each employee shall receive five hundred dollars (\$500.00) financial assistance for the purpose of clinical education. The amount of this financial assistance shall increase to five hundred and fifty dollars (\$550.00) effective April 1, 2006.

Employees will not be required to submit receipts.

### (e) ACP Leave of Absence

Upon request of an employee and subject to operational requirements, the Employer shall grant a leave of absence for the purpose of attending ACP school, in accordance with the following terms and conditions:

- (i) application: the employee shall make the request by submitting the Employer's "Education Request Form" and "Leave of Absence Request Form", together with proof of the employee's acceptance into the program for which leave is requested, all in accordance with the procedure set out in the Employer's policy on Educational Leave Assistance;
- (ii) approved program: the proposed paramedic education program must be CMA approved, unless waived by the Employer, and meet EHS standards; it is understood that distance education programs do not qualify for this leave:
- (iii) basis for selection: preference shall be given to applicants based on seniority provided that, in order to level the distribution of ACP employees throughout the Province, the Employer may give preference to applicants from a site where there is a relatively low percentage of ACP employees. Employees who are selected based on their site are required to remain at that site until commencement of the leave and then return to service at the site in accordance with clause (vii).
- (iv) financial assistance: the Employer will provide seven thousand dollars (\$7,000) in financial assistance to be used for program fees, for a maximum of twenty-five (25) employees per fiscal year; the amount of this financial assistance shall increase to eight thousand dollars (\$8,000) effective April 1, 2006;
- group insurance: while an employee is on leave attending ACP school, the Employer shall pay one hundred percent (100%) of the employee's group insurance premiums for medical, extended health and life (standard, not optional) insurance;
- (vi) service and seniority: while on such leave, service and seniority shall be deemed continuous and an employee shall continue to accrue and accumulate service and seniority credits, provided however service accumulated during such leave shall not be used for the purpose of calculating vacation leave credits and sick leave credits;
- (vii) return to service: employees who receive the financial assistance referred to in clause 21.05 (e)(iv) above are required to return to service with the Employer at the employee's regular site for fifteen (15) months following completion of ACP school - and will be required to sign the Employer's "Agreement for Return to Service" form before the Employer disburses

any financial assistance. If an employee fails to meet the service commitment, the employee is required to repay the financial assistance, prorated based on the amount of the unfulfilled commitment. (For example, an employee who only fulfills  $\sin(6)$  months of the fifteen (15) month commitment will be required to repay sixty percent (60%) of the financial assistance.) An employee may, at the discretion of the Employer, be released from his/her service commitment.

## 21.06 Application

For the purposes of this Article, "employee" means a permanent employee (that is, a Full-Time employee or Regular Part-Time employee).

## 21.07 Notice of Layoff

- (a) Eight (8) weeks notice of layoff shall be sent by the Employer to the Union and the employee(s) who is/are to be laid off, except where a greater period of notice is provided for under (b) below.
- (b) When the Employer lays off ten (10) or more persons within any period of four (4) weeks or less, notice of layoff shall be sent by the Employer to the Union and employees who are to be laid off, in accordance with the following:
  - (i) eight (8) weeks if ten (10) or more persons and fewer than one hundred (100) persons are to be laid off;
  - (ii) twelve (12) weeks if one hundred (100) or more persons and fewer than three hundred (300) are to be laid off:
  - (iii) sixteen (16) weeks if three hundred (300) or more persons are to be laid off.
- (c) Notices pursuant to this Section shall include the effective date of layoff and the reasons therefore.
- (d) An employee in receipt of lay off notice shall be entitled to exercise any of the following options:
  - to exercise placement/displacement rights in accordance with the procedure set out in this Article; or
  - (ii) to accept layoff and be entitled to recall in accordance with Article 21.11;or
  - (iii) to resign and accept severance pay in accordance with Article 32.01(a);

An employee who intends to exercise placement/displacement rights pursuant to (d)(i) above will indicate such intent to the Employer within five (5) full days following receipt of the layoff notice. If the employee does not indicate such intent within this period, he/she will be deemed to have opted to accept layoff in

accordance with (d)(ii) above.

#### 21.08 Pay in Lieu of Notice

Where the notice required by Article 21.07 is not given, the employee shall receive pay, in lieu thereof, for the amount of notice or balance thereof to which the employee is entitled

## 21.09 Placement

- (a) An employee in receipt of layoff notice may choose to accept placement within a vacant position, for which he/she is qualified as referenced in Article 21.10, within his/her administrative area or any region of the Province.
- (b) A permanent employee in receipt of layoff notice, who is placed in a term position, shall retain his/her status as a permanent employee.
- (c) An employee will have a maximum of two (2) full days to exercise his/her rights in any of the steps of the placement/displacement procedures provided for herein.
- (d) For a period of one (1) year, employees who have been transferred due to circumstances identified in Article 21, shall be eligible for transfer back to his/her former site, should a position become available provided he/she has the qualifications as referenced in Article 21.10. When two (2) or more employees request transfer back to a former site, the employee with the greatest length of seniority will have preference.
- (e) An employee who has been transferred to a site due to circumstances identified in Article 21 shall, notwithstanding the limitation in Article 20.03 (b)(i), be eligible to apply for any posted vacancy subject to the following:
  - (i) if the employee is a probationary employee at the time of transfer, the employee has completed his/her probationary period;
  - (ii) the employee has completed at least ninety (90) days of employment at the new site; and
  - (iii) if the employee is successful in any application for a posted vacancy, the employee is no longer eligible under Article 21.09 (d) for transfer back to the site from which he was transferred due to circumstances identified in Article 21.

Example: an employee at site "A" is transferred to site "B" due to circumstances identified in Article 21. After the employee has been employed at site "B" for at least ninety (90) days and provided the employee has completed his/her probationary period, the employee is eligible to apply for any posted vacancy at another site. If there is a posted vacancy for site "C" and the employee is successful in his/her application for that posting, the employee will no longer be eligible for transfer back

to site "A" under Article 21.09 (d).

# 21.10 <u>Displacement Procedures</u>

- (a) For the purpose of this Article, PCP (B.L.S.) is considered a single classification and, ICP and ACP (A.L.S.) is considered a single classification.
- (b) Employees in the BLS classification who are employees as of the signing of this Agreement will not be laid off for the purpose of creating an ALS position. Employees in the BLS classification who are hired after the signing of this Agreement, and who have completed two (2) years of service as a full-time employee, will not be laid off for the purpose of creating an ALS position.
- (c) Employees who have less than two (2) years of seniority may be laid off by classification within the site. The least senior person of those employees with less than two (2) years seniority within the classification selected for reduction shall be the person laid off first. The person displaced from a site has the option of:
  - (i) displacing either of the two most junior employees within his/her classification within the administrative area, unless the two (2) most junior employees within the administrative area are employed at the same site in which case, the employee with the least seniority shall be the one displaced; or
  - (ii) displacing the most junior employee within their classification within their region or the Province.

The person displaced within the region, if so chosen, shall then displace the most junior person within their classification within the Province.

(d) Employees with two (2) years or more seniority that are to be laid off shall be laid off in reverse order of seniority within a site regardless of classification.

Employees with two (2) years or more seniority who are laid off from a site have the option of displacing:

- (i) either of the two (2) most junior employees within the administrative area, regardless of classification, unless the two (2) most junior employees within the administrative area are employed at the same site, in which case the employee with the least seniority shall be the one displaced; or
- (ii) the most junior employee, regardless of classification, within either their region or within the Province.

The most junior employee displaced from a region, if he has less than two (2) years seniority, may elect to displace the most junior employee in the Province within his classification; or if he has two (2) or more years seniority, may elect to displace the most junior employee in the Province of either classification.

- (e) An employee who is displaced in accordance with this Article shall be entitled to the full rights contained in this Article.
- (f) An employee who chooses to exercise rights in accordance with Article 21.10 may elect at any step to accept layoff and to be placed on the recall list or to resign with severance pay.
- (g) An employee will have a maximum of two (2) full days to exercise his/her rights at any of the foregoing steps of the displacement procedures provided for herein.

# 21.11 Recall Procedures

Employees who are laid off shall be placed on the Recall List.

- (a) Employees will be recalled by seniority regardless of classification provided however employees who have less than two (2) years seniority at the time of layoff shall be recalled by seniority within their classification as defined in Article 21.10(a).
- (b) An employee shall be notified of the opportunity for recall by telephone, fax and/or in person. A formal verification in writing will be provided where the initial contact of recall is other than in writing.
- (c) The Employer shall notify an employee of recall by Registered Mail. If the employee fails to respond to the notice of recall with two (2) working days after receipt of the notice or fails to report to work within fourteen (14) calendar days after being advised of the recall, he/she shall be struck from the recall list and shall be deemed to have resigned unless he/she has notified the Supervisor that he/she is unable to return to work due to illness. This provision shall have no application in the case where an employee refuses recall for casual work or term employment of less than twelve (12) months if the employee is employed elsewhere.
- (d) Employees are responsible for leaving their current address and phone number with the Employer.
- (e) An employee on layoff shall supersede the call in list and shall be offered work on a casual basis within the Administrative Area of the site from which he/she was laid off, up to the point when he/she has worked the regular hours for that site. When the employee is working as a Casual, his/her status as a laid off employee shall not change; he/she is considered a Casual and therefore not covered by provisions of this Collective Agreement, except the applicable rate set out in Schedule A for the site where he/she is working as a Casual. The total of the days worked on a casual basis shall extend the recall period set out in Article 21.12 by that total.
- (f) Subject to (a) above, no new employees shall be hired within the bargaining unit until those laid off have been given the opportunity of recall.

(g) Where the Employer's business plan requires a site to be relocated, the affected employee(s) will move to the new site with full salary and benefits enjoyed at the original site, provided that if the relocation exceeds a radius of sixteen (16) kilometers for sites in Cape Breton Regional Municipality and Halifax-Dartmouth-Bedford-Sackville or a radius of thirty-two (32) kilometers in all other areas, the employee(s) may elect to treat the relocation as a layoff.

# 21.12 Termination of Recall Rights

If the layoff lasts for more than twelve (12) consecutive months without recall, the layoff shall be a termination of employment and recall rights shall lapse. Similarly, if an employee returns from a disability to the Recall List and remains on that list for more than twelve (12) consecutive months without recall, employment shall be terminated and recall rights shall lapse.

## **ARTICLE 22 - JOINT CONSULTATION**

- 22.01 Within sixty (60) days of the signing of the Collective Agreement, the parties agree to establish a Labour Management Committee in each administrative area comprised of equal representation to meet for the purpose of discussing matters not pertaining to the Collective Agreement.
- 22.02 The terms of reference for the Labour Management Committees shall be developed by the Joint Committee on Labour Relations in accordance with Article 21.01.
- 22.03 Unless otherwise mutually agreed upon, meetings of the Labour Management Committee(s) shall be scheduled at least quarterly. Either party may submit agenda items.

# ARTICLE 23 - HEALTH AND SAFETY

# 23.01 Occupational Health and Safety Act

The Employer agrees to be bound by the provisions of the *Occupational Health and Safety Act*, S.N.S. 1996, c7. Any breach of the Employer's obligations under that *Act* may be grieved pursuant to Article 11.

# 23.02 Occupational Health and Safety Committee(s)

In accordance with the *Occupational Health and Safety Act*, there shall be an Occupational Health and Safety Committee(s).

# 23.03 Alcohol. Drug and Gambling Dependency

Without detracting from the existing rights and obligations of the parties recognized in other provisions of this Agreement, the Employer and the Union agree to cooperate in encouraging employees afflicted with alcoholism, drug dependency or gambling

dependency, to undergo a coordinated program directed to the objective of their rehabilitation.

## ARTICLE 24 - UNIFORMS

# 24.01 Uniforms and Equipment

(a)

(b)

	FULL-TIME	REGULAR PART-TIME
Shirts	6	3
Pants	4	2
Socks	5	2
Black 2-1/2" belt	I	1
3 Seasons Jacket	1	1
Spring/Summer Jacket	1	1
Ball Cap	1	I
Winter Toque	1	1
1 Pair Safety Glasses	1	1

All new employees hired since the previous winter will be provided with a coat.

- (c) The Employer shall replace the above components of the uniform as reasonably required upon return of the worn out clothing, but an employee shall receive no more than two (2) pairs of socks per calendar year. The Employer will make reasonable efforts to supply and replace the uniforms in a timely manner.
- (d) The Employer will provide one (1) stethoscope for each ambulance.
- (e) The Employer shall reimburse employees who have completed their probationary period for the cost of purchase of components of the EHS approved standard uniform or standard paramedic equipment, to a maximum of fifty dollars (\$50.00) per year, upon presentation of receipt(s). The maximum will increase to one hundred dollars (\$100.00) per year effective April 1, 2006.
- (f) Upon termination of employment, an employee is required to return all shirts, vests, jackets and ball caps provided as parts of the uniform. If an employee fails to do so, the employee is liable to the Employer for the cost of any part of the uniform issued to the employee. The Employer is authorized to deduct such

amount from the employee's pay, including vacation pay, and after any such deduction, the employee is then liable for any unpaid balance.

#### 24.02 Safety Boots

Employees are required to wear CSA approved safety boots as specified in the EHS Uniform Standards, as may be amended from time to time. The Employer shall reimburse employees who have completed their probationary period for the cost of the purchase of CSA approved safety boots to a maximum of one hundred and fifty dollars (\$150.00), upon presentation of a receipt. The Employer shall reimburse employees for the cost of replacing safety boots to a maximum of one hundred fifty dollars (\$150.00) every two (2) years, upon presentation of a receipt; provided however, in exceptional circumstances, the Employer will consider such reimbursement before the expiry of two (2) years. The boot allowance will be increased to one hundred and eighty dollars (\$180.00) effective April 1, 2006.

# **ARTICLE 25 - GENERAL CONDITIONS**

# 25.01 Rate of Pay During Accident Investigation

All employees covered by this Agreement who, while operating vehicles or equipment maintained by the Employer, become involved in any collision or accident with the said vehicle or mobile equipment, will continue to receive their normal rate of pay until the investigation of the collision or accident has been completed by the Employer.

# 25.02 Crew Ouarters

- (a) The Employer agrees to make the following available to employees at crew quarters:
  - (i) Cable TV the Employer will pay for the cable hookup or, where costs of cable hookup are prohibitive, a satellite dish hookup, to a maximum of one hundred dollars (\$100.00) per site, provided employees based at the site pay the monthly charge;
  - (ii) food preparation area;
  - (iii) fridge and microwave or stove; (existing sites which have both a microwave and stove shall continue to have both microwave and stove);
  - (iv) toilet facilities;
  - (v) personal lockers;
  - (vi) adequate furniture sofa, chairs, tables, desk;
  - (vii) appropriate ventilation equipment;
  - (viii) sleeping quarters and beds (only at sites where there are 24 hour shifts);

- (ix) dishes, pots, silverware;
- (x) potable drinking water (if the tap water is potable, the Employer will pay for a cooler if employees based at the site pay for bottled water).
- (b) Employees shall have reasonable access to showers either at their own crew quarters or nearby (not in a private home).
- (c) The Employer shall arrange for professional cleaning of each base as follows: floors and carpets to be cleaned on a quarterly basis; walls and windows to be cleaned on an annual basis.

#### 25.03 Driver's License

Each employee shall be required to:

- (a) advise the Employer as soon as reasonably possible if the license allowing him/her to drive an ambulance (presently Class 4 or equivalent) is suspended or revoked or if there are any charges (including both charges incurred while driving an ambulance and charges incurred while driving a vehicle other than an ambulance, for example, charges for speeding in a personal vehicle) which may either immediately or in the future affect the validity of the driver's license; and
- (b) upon the Employer's request, provide a driver's abstract, with any costs incurred to be at the Employer's expense.

# 25.04 EHS Registration

**All** employees must, as a condition of employment, be registered with the Province's Minister of Health at the PCP, ICP or ACP level. It is understood that the Employer will adhere to all rulings by EHS regarding registration.

## 25.05 Deep Cleaning of Vehicles

Employees who are required to clean vehicles are not required to do the following work on the exterior of the vehicle: waxing, detarring or tire dressing.

## ARTICLE 26 - PAY PROVISION

## 26.01 Rate of Pay

The rates of compensation for employees shall be as contained in Schedule A which shall form part of this Agreement.

# 26.02 Rate of Pay upon Appointment

Subject to Article 26.03, the rate of compensation of a person upon appointment to a position shall be the minimum rate prescribed for the classification to which he/she is appointed.

SMSS\\1051363 | 15

# 26.03 Exception

The rate of compensation of an employee upon appointment may be at a rate higher than the minimum rate prescribed for the classification, if, in the opinion of the Employer, such higher rate is necessary to effect the appointment of a qualified person to the position or if the person to be appointed to the position has qualifications in excess of the minimum requirements for the position.

### 26.04 Rate of Pay upon Reclassification

The rate of compensation of an employee upon progression to the ACP classification shall be at a rate which is the greater of the minimum for an ACP or a rate which gives the employee an increase of at least \$1.00 per hour more than the rate received by the employee before the progression to ACP.

## 26.05 Anniversary Date

The anniversary date of an employee shall be the first day of the month in which employment occurs if the employee reported for duty during the first fifteen (15) calendar days of the month in which he/she was employed, or the first day of the following month if the employee reported for duty later than the fifteenth (15th) calendar day of the month. The anniversary date will only change to the first day of another month if:

- (a) the employee is reclassified, at which time the date of the reclassification becomes his/her new anniversary date; or
- (b) the employee has been on leave of absence without pay, in which case the employee's anniversary date will be moved forward by the amount of time which the employee was on leave without pay, unless otherwise provided in this Agreement.

# 26.06 Salary Increments

The Employer, except as provided for in Article 26.07, may grant an increment for meritorious service after an employee has served for a period of twelve (12) months following the first day of the month established in Article 26.05 or twelve (12) months following the date of a change in his/her rate of compensation as established in Articles 26.02 or 26.04.

# 26.07 Notice of Withheld Increment

When an increase provided for in Article 26.06 is withheld, the reason for withholding shall be given to the employee in writing by the Employer.

# 26.08 Granting of Withheld Increment

When an increase provided for in Article 26.06 is withheld, the increase may be granted on any subsequent first day of any month after the anniversary date upon which the increase was withheld.

#### 26.09 Premium for Senior Operations Paramedics

Employees designated as Senior Operations Paramedics (which includes employees sometimes referred to as Crew Chiefs) shall receive the following premiums above the higher rate of pay for the classification: \$2,20 per hour.

## 26.10 Rate of Pay other than Regular Rate

There are circumstances where an employee will be paid at an hourly rate different than his/her regular hourly rate:

- (a) if an employee is working as a Casual while on layoff (see Article 21.11 (e)), he/she shall be paid not less than the hourly rate that he or she was receiving at the time of layoff; and
- (b) if an employee is temporarily transferred by the Employer under Article 20.05, the employee shall be paid in accordance with that Article.

## ARTICLE 27 · HOURS OF WORK

# 27.01 No Guarantee of Hours

An employee's regularly scheduled hours of work shall not be construed as guaranteeing the employee minimum or maximum hours of work, but is the basis for computing compensation for hours in excess of regularly scheduled hours.

## 27.02 Regular Work Week

The regular work week for a Full-Time employee shall, on average, consist of forty-two (42) hours per week.

#### 27.03 Shifts

- (a) Subject to paragraphs (b) through (d), shifts shall be scheduled according to operational requirements and consistent with efficient operations. The Employer reserves the right to rotate shifts or to have employees work a combination of days and nights.
- (b) Regularly scheduled shifts shall not be for less than eight (8) hours and there shall be no split shifts.
- (c) There will be at least ten (10) hours rest between regularly scheduled shifts.
- (d) The Employer shall not change an employee's regularly scheduled shifts without:
  - (i) consultation with the Union prior to making any change; and
  - (ii) at least sixty (60) days' notice of the change to the employee(s) affected.

#### 27.04 Breaks and Allowances

(a) Employees shall be entitled to a one-half (½) hour paid break without interruption. An employee who is working throughout the hours set out below and who either does not have a reasonable opportunity to take a break within the hours or has his/her break interrupted, shall receive an allowance as follows:

\$8.00	11:00 a.m. to 3:30 p.m. (day time break)
\$12.00	5:00 p.m. to 9:00 p.m.
\$8.00	11:00 p.m. to 3:00 a.m.
\$8.00	5:00 a.m. to 9:00 a.m.

The above hours are based on a shift starting at 7 a.m. and hours as defined above shall be adjusted in accordance with different start times, the first hours beginning three and one-half (3-1/2) hours after the commencement of each shift.

It is understood that an employee must work throughout the full period of the hours to be entitled to a break. For example, an employee working the twelve (12) hour shift from 7 a.m. to 7 p.m. would only be entitled to the one (1) break during his/her shift - between 11 a.m. and 3:30 p.m.

Employees in the Halifax, Dartmouth, Bedford, Sackville and Cape Breton Regional Municipality shall be assigned day time breaks (but not other breaks) by the Employer through the Communications Centre.

- (b) Employees are entitled to spend their one-half (½) hour paid break at a base or posting with lunchroom facilities. Employees who are unable to do so shall receive the appropriate allowance set out in paragraph (a).
- (c) When an employee does not receive the required break, or does not have an opportunity to take the break, in addition to paying the employee the allowance set out in paragraph (a) above, the Employer shall make every reasonable effort to arrange for the employee to receive a one-half (½) hour break later in the shift. If the employee is not given an opportunity to take the break later in the shift, the employee shall be paid an additional allowance of Five Dollars (\$5.00) for the shift.
- (d) When an employee performs a "long haul" trip with a distance, one way, of Two Hundred (200) kilometers or more, the employee shall receive an additional allowance of Five Dollars (\$5.00) per trip.
- (e) The Employer may grant allowances for other than the above where the Employer deems circumstances warrant.

# 27.05 Posted Schedule

The regular work schedule for all employees shall be posted in an appropriate place at least four (4) calendar weeks in advance of the schedule to be worked.

## 27.06 Compensation for Partial Shift

An employee who reports to work for his/her shift or any other shift and is sent home due to lack of work shall be compensated for one full shift.

## 27.07 Consecutive Days Off

Each employee shall be scheduled for minimum of two (2) consecutive days off in each week unless other acceptable arrangements are mutually agreed upon.

## 27.08 Stand-By

- (a) Stand-by is a condition of employment whereby employees are required and so designated by the Employer to maintain themselves immediately available for work during a defined period outside their regular hours of work.
- (b) Employees shall only be required to be on stand-by in the event of an emergency situation of wide scope such as Y2K, the crash of Swiss Air Flight 111 or "9/11" (the diversion of air traffic in Nova Scotia on September 11, 2001. Employees shall not be required to be on stand-by for more than twenty-four (24) hours per week above their regularly scheduled hours.
- (c) Employees on stand-by shall be required to carry a pager and the Employer shall pay all costs associated with the pager.
- (d) An employee shall be paid Two Dollars and Fifty Cents (\$2.50) per hour for each hour he/she is scheduled to be on stand-by.
- (e) An employee on stand-by is required to report for duty when called and will be paid the applicable regular hourly rate upon reporting for duty. When reporting for duty on a stand-by, the employee will be guaranteed a minimum of four (4) hours regular pay.

## ARTICLE 28 - OVERTIME

# 28.01 Overtime

Overtime means work in excess of an employee's regularly scheduled hours as set out in Article 27. Overtime may or may not result in an employee being compensated at a higher rate than his/her regular hourly rate.

## 28.02 Call-in System

- (a) The Employer shall offer extra shifts on a voluntary basis to employees in accordance with the call-in system set out in Article 28.08. Employees who volunteer to work the extra shifts shall be compensated at the straight time rates which apply at the site of the call-in.
- (b) If the Employer is unable to obtain sufficient volunteers at straight time rates \$MSSM105136318

under paragraph (a), the Employer is unrestricted in assigning the available shifts to casuals, supervisors, or other person excluded from the bargaining unit so long as the assigned hours to any non-bargaining unit employee do not exceed forty-eight (48) hours in any week.

For greater certainty, supervisors or non-bargaining employees shall not be assigned or scheduled to work overtime hours, i.e. hours in excess of forty-eight (48) in any week, unless and until those hours have first been offered to all bargaining unit employees who have indicated their availability to work overtime hours on designated shifts.

For further certainty, a casual, supervisor or non-bargaining employee who has commenced working on a scheduled shift in conformity with this article 28.02 (b) and as a result of operational requirements arising after the start of a shift is required to continue working beyond the regularly scheduled shift, may continue to work to the end of the shift overrun even though this may cause the employee to exceed 48 hours work in a week.

(c) If the Employer is still unable to fill the shifts and wishes to do so, then the Employer shall offer the shifts to employees in accordance with a call-in system set out in Article 28.08, but at premium rates, that is time and one-half  $(1\frac{1}{2}x)$  the rates for the site of the call-in.

#### 28.03 Premium Pay

An employee shall be compensated at the rate of time and one-half  $(1\frac{1}{2}x)$  the applicable regular hourly rate for time worked where an employee:

- is required to work in excess of his/her regularly scheduled hours at the end of his/her regular shift or beyond the end of a call-in shift, in both cases subject to Article 28.04 (Overtime Eligibility); or
- (b) works an extra shift offered to the employee in the second round of the call-in system referred to in Article 28.02(c).

# 28.04 Overtime Eligibility

An employee must work at least fifteen (15) minutes beyond his/her regular shift before being eligible for premium pay.

# 28.05 Computation of Overtime

In computing overtime, a period of thirty (30) minutes or less shall be counted as one-half (½) hour and a period of more than thirty (30) minutes but less than sixty (60) minutes shall be counted as one (1) hour.

#### 28.06 Form of Compensation/Banking Excess Hours

Compensation for overtime shall be paid except where, upon request of the employee, overtime will be granted in the form of time off in lieu of overtime hours worked.

- (a) Where an employee works hours in excess of his regularly scheduled hours of work, the employee may bank the excess hours, along with any hours banked under Article 19.04 (work on a holiday).
- (b) Employees may request time off with pay from the banked hours and may, subject to operational efficiency, be granted time off with pay drawn from his/her banked hours.
- (c) When a Full-Time Employee is regularly scheduled for only 72 hours or less in a pay period, the employee may draw a block of 12 hours from his banked hours, subject to the following qualifications:
  - (i) if the employee has less than 12 hours banked, the employee may withdraw the remaining hours in his/her bank (but cannot create a negative balance); and
  - (ii) withdrawal of banked hours cannot result in an employee receiving more pay than if he/she had worked 84 hours in the pay period.

This clause (c) shall only apply on elimination of "pay smoothing"

- (d) Banked hours remaining in the bank as of the end of the first full pay period in November in any year shall be paid out on the first pay in December.
- (e) In all cases, banked hours will be paid out at the rate they were earned with the hours first banked to be paid out first (first in, first out).
- (f) It is understood that the determination of whether operational efficiency permits time off to be granted shall be at the sole and absolute discretion of the Employer (and during the life of this Collective Agreement shall not be the subject of a grievance). As a means of resolving chronic on-going difficulties in granting time off at a particular site, the issue may be referred to the Joint Committee on Labour Relations.

# 28.07 Call In

An employee required to report back to work after leaving the premises of the work location following completion of a shift but before the commencement of his/her next shift or called in to work on a day the employee is not scheduled to work, shall be granted a minimum of four (4) hours at the straight time rate or the applicable overtime rate, whichever is greater.

#### 28.08 Call-In System

## (a) Availability Lists:

- (i) Availability lists will be used to indicate an employee's willingness to work hours outside the employee's regularly scheduled hours at individual sites within the employee's administrative area.
- (ii) An employee who places his/her name on availability list shall provide the Employer with at least one (1) contact number (telephone number or pager number), but no more than two (2) contact numbers.
- (iii) Employees may indicate, in writing, their willingness to be either added to, or deleted from, individual sites on the availability list. Any such changes will be effective the first business day of the month after the employee has given the Employer at least twenty-four (24) hours' notice of the change.
- (iv) In administering the call-in system and subject to the exception set out in Article 28.08 (f), the Employer is not obligated to call employees for sites for which an employee has not indicated on the availability list that he/she is willing to work.

## (b) Call-In Lists

- (i) The call-in system will be based on call-in lists which will be prepared both by site and by administrative area, in both cases by classification, that is Basic Life Support (BLS) and Advanced Life Support (ALS); (for the purposes of this provision, ICP and ACP shall be considered to be the ALS classification). As a result, for each site, there will be a BLS call-in list and an ALS call-in list ("site BLS list" and "site ALS list"); for each administrative area, there will be a BLS call-in list and an ALS call-in list ("administrative area BLS list" and "administrative area ALS list").
- (ii) For the purposes of this Article 28.08 (call-in system), the Halifax-Dartmouth-Bedford-Sackville administrative area shall be considered one (1) site.
- (iii) Employees are only eligible for the site call-in list for their regular site and for the administrative area call-in list for the administrative area which includes their regular site.
- (iv) Where two (2) contact numbers have been provided by an employee, both shall be called by the person(s) administering a call-in. Where there is a message service, a message will be left for the employee advising him/her of the date(s) or day(s) of the available shift(s).
- (v) An employee who accepts a call-in shall be moved to the bottom of the call-in list from which he/she was called. (If an employee accepts a call-in

from the site call-in list, he/she moves to the bottom of that list, but does not move on the administrative area call-in list.) An employee who refuses a call-in shall not be moved. If an employee is added to the call-in lists, the Employer shall place the employee on the call-in lists (both site and administrative area) by starting at the top of each list and placing him immediately above the first employee on the list who is junior to the employee being added.

# (c) Less than Six (6) Hours' Notice

Where the Employer has less than six (6) hours' notice of the need for a call-in, the Employer is unrestricted in offering the shifts.

# (d) Six (6) Hours or More. But Less Than Twenty-Four (24) Hours' Notice

- (i) Where the Employer has six (6) hours or more, but less than twenty-four (24) hours' notice, of the need for a call-in, the Employer shall administer the call-in system based on the site call-in lists only, first by the classification being replaced (BLS/ALS) and then by the other classification. For example, if a BLS employee is being replaced, persons are called in first from the site BLS list and then from the site ALS list; conversely, if an ALS employee is being replaced, persons are called in first from the site ALS list and then from the site BLS list; there is no requirement to proceed to the administrative area call-in lists. If no one on the site call-in lists is available or can be contacted at the time of the call, then the Employer is unrestricted in offering the shifts.
- (ii) If the Employer is unable to contact an employee after calling the contact number(s) provided by the employee, but is able to leave a message for the employee requesting the employee to call, if the employee responds within fifteen (15) minutes, he/she will be awarded the shift. Prior to expiry of the fifteen (15) minutes, the Employer may proceed to the next names on the call-in list, but any other employee shall be offered the shift subject to the employee higher on the call-in list with whom the message was left returning the call and agreeing to take the shift during his/her allowed fifteen (15) minutes.

For example, if the Employer attempts to contact employee number one, but is unable to contact the employee and leaves a message for the employee to call and during employee number one's fifteen (15) minute response time, the Employer contacts employee number two who is lower on the call-in list, employee number two is to be awarded the shift only on condition that employee number one does not return the call and accept the shift during his/her fifteen (15) minute response time. If employee number one responds and accepts the shift during his/her fifteen (15) minute response time, the shift is assigned to employee number one. If employee number one does not respond and accept the shift during his/her fifteen (15) minute response time, that shift is unconditionally assigned to employee number two (even if employee number one subsequently

responds and wants to work the shift).

# (e) More Than Twenty-Four (241 Hours' Notice and ut) to Seven (7) Days' Notice

- (i) Where the Employer has more than twenty-four (24) hours' notice, and **up** to seven (7) days' notice, of the need for a call-in, the Employer shall administer the call-in system based on the call-in lists as follows:
  - (A) first, by the site call-in list for the classification being replaced;
  - (B) secondly, by the administrative area call-in list for the classification being replaced;
  - (C) thirdly, by the site call-in list for the other classification;
  - (D) fourthly, by the administrative area call-in list for the other classification.

For example, assuming a BLS is being replaced: first by site BLS list, secondly by administrative area BLS list; thirdly, by site ALS list and fourthly, by administrative area ALS list.

Conversely, if an ALS is being replaced: firstly by site ALS list; secondly, by administrative area ALS list; thirdly by site BLS list and fourthly by administrative area BLS list.

# (f) More Than Seven (7) Days' Notice

Where the Employer has more than seven (7) days' notice of the need for a call-in, the Employer shall post a notice for the available shifts for five (5) days so as to give employees an opportunity to indicate their willingness to work the shifts. Supervisors shall then administer the call-in system in accordance with paragraph (e), but instead of contacting employees individually, may use the response to the notice as a basis for assigning the work. To be eligible for call-in under this paragraph (f), it is not necessary for an employee to have indicated willingness to work at a site on the availability list under paragraph (a).

# (g) Separate Call-In Lists for Special Events and for Hospitals

- (i) The Employer may establish separate call-in lists by classification (ALS/BLS) in each administrative area for special events in that administrative area and also separate call-in lists in each administrative area for work in hospitals in that administrative area.
- (ii) Such call-in lists shall initially be arranged in order of seniority and shall then normally be operated on the same principles as the regular call-in lists, for example, an employee who accepts a call-in shall be moved to the bottom of the call-in list from which he/she was called, but an employee who refuses the call-in shall not be moved.

- (iii) Notwithstanding the general principles for operation of the separate call-in lists set out above:
  - A. an employee will not be offered a call-in until he/she has received any required orientation; for example, orientation is generally required for hospital work;
  - B. all assignments are subject to the requirements of the contractor; for example, normally hospitals request ALS and then BLS and normally for special events there is one ALS and one BLS, but a hospital may request an ACP instead of ALS, or a contractor for a special event may request two ALS or two BLS and call-ins will then be offered in accordance with the contractor's request.

## (h) Cancellation of Call-In

Where an employee accepts the call-in (and is therefore moved to the bottom of the call-in list), but the call-in is cancelled:

- (i) the employee will be placed at the top of the call-in list at the time of the cancellation; and
- (ii) when the cancellation occurs less than twenty-four (24) hours before the start of the cancelled shift, the employee shall receive four (4) hours' pay at his regular rate of pay.

The above provisions only apply where the reason for the cancellation is caused by the Employer (that is, "Employer driven") and also where an employee has been scheduled to attend in Court for circumstances arising out of his/her employment and the shift is cancelled because his/her attendance in Court is no longer required.

# (i) No Restriction on Use of Term Employees

It is understood that the call-in system does not restrict the Employer from using a Term employee (who is hired for and regularly is scheduled to work for a term in excess of three (3) months but not more than twenty-four (24) months).

# (j) Full Shifts or Half Shifts

- (i) In the case of a twenty-four (24) hour shift, an eighteen (18) hour shift or a sixteen (16) hour shift, the Employer has the option of attempting to fill the full shift or creating "half shifts", i.e. replacing a twenty-four (24) hour shift with two (2) twelve (12) hour shifts, replacing an eighteen (18) hour shift with two (2) nine (9) hour shifts and replacing a sixteen (16) hour shift with two (2) eight (8) hour shifts.
- (ii) The appropriate process to be followed for a call-in for the second half of any such "half shift" shall be determined at the time of the employee's

initial notification with respect to the original shift, not based on any follow-up call from the employee with respect to the possibility of filling the second half of the original shift.

## 28.09 Overtime Meal Allowance

An employee who is required to work a minimum of three (3) hours overtime following his/her scheduled shift shall be paid a meal allowance of \$10.00 in addition to the overtime rate of pay. In such circumstances, Article 27.04 shall not apply.

#### 28.10 Overtime Restriction

Except in the case of a disaster or extreme emergency, employees who are scheduled for twelve (12) hour shifts or less shall not work or be required to work in excess of twenty-four (24) continuous hours. Employees who are scheduled to work twenty-four (24) hour shifts shall not work or be required to work in excess of thirty-six (36) continuous hours.

# ARTICLE 29 - LEAVES OF ABSENCE

#### 29.01 Personal Leave

Employees may request a leave of absence without pay, provided such application is made in writing and the reason for such leave of absence is set out in such written request. Granting of personal leave shall be at the Employer's sole discretion.

#### 29.02 Sick Leave

(a) An employee who is unable to perform his/her duties on his/her regularly scheduled shift because of illness or injury, which is not compensable under WCB, for a period not exceeding the elimination period required under Weekly Indemnity may be granted leave with pay, not to exceed One hundred and forty-four (144) hours for employees who work eighty four (84) hours bi-weekly.

## (b) Proof of Illness

- (i) An employee who is ill shall notify the Employer as soon as reasonably possible upon recognizing that he/she will be unable to report for work.
- (ii) An employee may be required by the Employer to produce a certificate from a legally qualified medical practitioner for any period of absence exceeding one (1) shift for which sick leave is claimed by an employee and if a certificate is not produced after such a request, the time absent from work will be deducted from the employee's pay. Where the Employer has reason to believe an employee is misusing sick leave privileges, the Employer may issue to the employee a standing directive that requires the employee to submit a medical certificate for any period of absence for which sick leave is claimed.

- (iii) For the purposes of proof of illness (and also for medical clearance to return to work), the Employer may require that the employee be examined by an alternate physician. If the employee is dissatisfied with the alternate physician selected by the Employer, the employee shall advise the Employer accordingly, in which case the Employer will provide the employee with the names of three (3) other physicians and the employee will select one (1) of the three (3).
- (iv) Where the Employer requires a medical certificate under clause (ii) or refers an employee to an alternate physician pursuant to clause (iii), and where medical fees in excess of those covered by any medical insurance are incurred by the employee, the Employer shall pay the cost of those fees.

#### 29.03 Other Uses for Sick Leave

An employee may also use their sick leave allowance for the following purpose:

# (a) Sick Leave for Immediate Family Illness

In the case of illness of a member of an employee's immediate family, meaning spouse, son, daughter, father or mother and when no one at home other than the employee can provide for the needs of the ill person, the employee may be granted, after notifying his/her immediate management supervisor, leave with pay up to one (1) working day per fiscal year, for the purpose of making such arrangements as are necessary to permit the employee's return to work. (This leave is to be taken only after an employee has used leave for family illness under Article 29.05.)

# (b) Leave for Medical and Dental Appointments

Employees shall be allowed paid leave of absence up to two (2) working days per fiscal year, in order to engage in personal preventative medical and dental care, subject to providing as much notice as reasonably possible to the Employer, and subject to operational requirements. Leave shall not be unreasonably refused; for example, the employer shall give consideration to whether the appointment is with a specialist and/or for an emergency.

# (c) Leave for Birth or Adoption of Child

An employee shall be granted leave with pay of one (1) shift either;

- (i) where an employee's spouse gives birth to a child; (and in the event of special circumstances such as a lengthy delivery, the leave may be extended into the employee's next scheduled shift, subject to a maximum of twenty-four (24) consecutive hours in total); or
- (ii) for the purpose of adoption of a child pursuant to the laws of the Province.

## (d) Leave for Emergency

An employee shall be granted leave of absence with pay of one (1) shift per fiscal year to attend to an emergency of a nature not provided elsewhere in this Agreement which cannot be attended to by others or attended to by an employee at a time when he/she is normally off duty. In the event of special circumstances, the leave may be extended into the employee's next scheduled shift, subject to a maximum of twenty-four (24) hours in total.

# 29.04 Summary/Clarification

For clarification, the combined use of sick leave shall not exceed the number of hours set out in Article 29.02, and within that time:

- (a) Leave for personal illness or injury shall not exceed the number of hours set out in Article 29.02;
- (b) Sick leave for family illness shall not exceed one (1) day per fiscal year;
- (c) Leave for medical or dental appointments shall not exceed two (2) working days per fiscal year;
- (d) Leave for the birth of a child or adoption shall not exceed the number of shifts or hours set out in Article 29.03 (c);
- (e) Leave for emergency shall not exceed the number of shifts and hours referred to in Article 29.03 (d).

Employees who exhaust all or part of their allotted sick leave entitlement in one (1) fiscal year shall have it reinstated on April 1 of the following fiscal year.

# 29.05 Leave for Family Illness

In the case of illness of an employee's spouse, son, daughter, father, mother or legal guardian and when no one at home other than the employee can provide for the needs of the ill person, the employee may be granted, after notifying his immediate management supervisor, leave with pay up to two (2) working days per fiscal year, for the purpose of making such arrangements as are necessary to permit the employee's return to work. (This leave is to be used before any leave for family illness under Article 29.03 (a).)

## 29.06 Proration of Sick Leave

- (a) A new employee who is appointed subsequent to April 1 shall have his/her maximum sick leave entitlement for the first fiscal year prorated in accordance with the number of months of service he/she will accumulate in the fiscal year of appointment.
- (b) Where, as a result of a job posting, an employee is transferred from one site to another and the transfer results in a change in the employee's regular weekly hours of work (and therefore the amount of sick leave), the employee's sick leave

entitlement for the year shall be prorated based on the amount of time during the year spent at each site.

# 29.07 Court Leave

- (a) Any employee who is required by subpoena or summons to attend as a witness in any Court proceeding which is work related shall be entitled to the following:
  - (i) when such attendance is required at a time when the employee would otherwise be at work leave of absence without loss of regular pay:
  - (ii) when such attendance is required at a time when the employee is scheduled off or on leave pay for time in Court, plus reasonable travel time, all subject to a maximum of twelve (12) hours per day and a minimum of four (4) hours per day.
- (b) Any employee who is required by subpoena to attend as a witness in an arbitration under this Collective Agreement shall be granted leave of absence with regular pay provided that where an employee is not subpoenaed by the Employer, the Union shall reimburse the Employer for fifty percent (50%) of the amount paid to the employee for the leave.

When such attendance is required by the Employer at a time when the employee is scheduled off or on leave - pay for time in arbitration, plus reasonable travel time will be paid to a maximum of twelve (12) hours per day and a minimum of four (4) hours per day.

## 29.08 Pregnancy Leave

- (a) <u>Pregnancy leave</u>: A pregnant employee, who has been employed by the Employer for at least one (I) year, is entitled to an unpaid leave of absence of **up** to seventeen (17) weeks upon
  - giving the Employer notice of the date that she will begin the leave and the date she will return to work as required by Article 29.08 (f), and
  - (ii) providing to the Employer, where the Employer so requests, a certificate of a legally qualified medical practitioner stating that the employee is pregnant and specifying the expected date of delivery.
- (b) <u>Beginning of leave</u>: Pregnancy leave pursuant to this Article begins on such date, not sooner than sixteen (16) weeks preceding the expected date of delivery, as the employee determines, and not later than the date of delivery.
- (c) End of leave: Pregnancy leave pursuant to this Article ends on such date
  - (i) not sooner than one (1) week after the date of delivery; and
  - (ii) not later than seventeen (17) weeks after the pregnancy leave began pursuant to this Article,

SMSS\\105136315

- as determined by the employee.
- (d) Requirement by Employer to take leave: Notwithstanding Article 29.08 (a), (b) and (c), the Employer may require a pregnant employee, who has been employed by the Employer for at least one (1) year, to take an unpaid leave of absence while the duties of her position cannot reasonably be performed by a pregnant woman or the performance of the employee's work is materially affected.
- (e) <u>Human Rights Act</u>: For greater certainty, nothing in this Article affects any protection provided to a pregnant employee regardless of the length of employment, by the Nova Scotia *Human Rights Act*.
- (f) Notice: An employee shall give the Employer four (4) weeks' notice of
  - (i) the date the employee will begin pregnancy leave pursuant to this Article; and
  - (ii) the date the employee will return to work upon completion of the leave unless the employee will take the maximum leave to which the employee is entitled
- (g) Amendment of notice: The notice given pursuant to Article 29.08 (f) may be amended from time to time by the employee:
  - (i) by changing any date in the notice to an earlier date if the notice is amended at least four (4) weeks before that earlier date;
  - (ii) by changing any date in the notice to a later date if the notice is amended at least four (4) weeks before the original date; and
  - (iii) by adding the date that the employee will return to work if the notice is amended at least four (4) weeks before the employee would have been required to return to work.
- (h) Short notice: An employee shall give the Employer through the Director of Human Resources as much notice as reasonably practicable of:
  - (i) the date the employee will begin the pregnancy leave where she is advised by a legally qualified medical practitioner to begin the pregnancy leave sooner than planned because of medical circumstances resulting from her pregnancy; and
  - (ii) the delivery where the actual delivery occurs sooner than expected; and Article 29.08 (f) does not apply.
- Notice in Writing: Notice given pursuant to this Article shall be put in writing
  where the Employer so requests.

- (j) Resumption of work: When an employee returns to work upon the expiry of a leave of absence taken pursuant to this Article 29.08, the Employer shall permit the employee to resume work in the same position she held prior to the commencement of the pregnancy leave, with no loss of seniority or benefits accrued to the commencement of the pregnancy leave. Where the position no longer exists, the matter shall be referred to the Joint Committee on Labour Relations.
- (k) <u>Maintenance of benefit plan</u>: While an employee is on pregnancy leave, the Employer shall maintain coverage for Group Life (Article 31.02), Extended Health Benefit Plan (Article 31.03), Weekly Indemnity and Long Term Disability (Article 31.04) and shall pay the premium cost for maintaining such coverage during the period of the pregnancy leave.
- (I) Service and seniority: While on pregnancy leave, an employee shall continue to accrue and accumulate service and seniority credits for the duration of her leave, and her service and seniority shall be deemed to be continuous. However, service accumulated during pregnancy leave shall not be used for the purposes of calculating vacation leave credits. Vacation leave credits will be prorated for the period of time the employee is not on such leave during the vacation year.
- (m) <u>Illness during pregnancy</u>: Leave for illness of an employee arising out of or associated with her pregnancy prior to the commencement of, or the ending of, pregnancy leave granted in accordance with Article 29.08 may be granted in accordance with the provisions of Article 31.04 (Weekly Indemnity).

## 29.09 Pregnancy Leave Allowance

The Employer agrees to pay a pregnancy leave allowance effective on the date of signing of this Agreement in accordance with the following:

- (a) An employee entitled to pregnancy leave under the provisions of this Agreement, who provides the Employer with proof that she has applied for and is eligible to receive employment insurance (EI) benefits pursuant to the *Employment Insurance Act*, 1996, shall be paid an allowance in accordance with the Supplementary Unemployment Benefit (SUB) Plan.
- (b) In respect to the period of pregnancy leave, payment made according to the SUB Plan will consist of the following:
  - (i) where the employee is subject to a waiting period of two (2) weeks before receiving EI benefits, payments equivalent to seventy-five percent (75%) of her weekly rate of pay for each week of the two (2) week waiting period, less any other earnings received by the employee during the benefit period; and
  - (ii) **up** to a maximum of fifteen **(15)** additional weeks, payments equivalent to the difference between the weekly EI benefits the employee is eligible to receive and eighty percent (80%) of her weekly rate of pay, less any other

earnings received by the employee during the benefit period which may result in a decrease in the EI benefits to which the employee would have been eligible if no other earnings had been received during the period.

- (c) For the purposes of this allowance, an employee's weekly rate of pay will be one-half the bi-weekly rate of pay to which the employee is entitled for her classification on the day immediately preceding the commencement of her pregnancy leave. In the case of a part-time employee, such weekly rate of pay will be multiplied by the fraction obtained from dividing the employee's time worked (as defined for the purpose of accumulating service) averaged over the preceding twenty-six (26) weeks by the regularly scheduled full-time hours of work for the employee's classification.
- (d) Where an employee becomes eligible for a salary increment or pay increase during the benefit period, payments under the SUB Plan will be adjusted accordingly.
- (e) The Employer will not reimburse the employee for any amount she is required to remit to Human Resources Development Canada where her annual income exceeds one and one-half (1 ½) times the maximum yearly insurable earnings under the Employment Insurance Act.

#### 29.10 Parental Leave

- (a) Parental leave: An employee, who has been employed by the Employer for at least one (1) year, and who becomes, before or after this Article comes into force, a parent of one (1) or more children through the birth of a child or children is entitled to an unpaid leave of absence of, subject to Article 29.10 (d), up to fifty-two (52) weeks upon giving the Employer notice of the date that the employee will begin the leave and the date the employee will return to work, as required by Article 29.10 (g), (h) and (i).
- (b) <u>Beginning and end of leave after pregnancy leave</u>: Where an employee takes pregnancy leave pursuant to Article 29.08 and the employee's new-born child or children arrive in the employee's home during the pregnancy leave, parental leave pursuant to this Article:
  - (i) begins immediately upon completion of the pregnancy leave and without the employee returning to work; and
  - (ii) ends not later than thirty-five (35) weeks after the parental leave began pursuant to this Article, as determined by the employee.
- (c) <u>Beginning and end of leave otherwise</u>: Where Article 29.10 (b) does not apply, parental leave pursuant to this Article
  - (i) begins on such date, coinciding with or after the birth of the child or children or the child or children first arriving in the employee's home; and

(ii) ends not later than fifty-two (52) weeks after the child or children first arrive in the employee's home,

as determined by the employee.

- (d) <u>Maximum leave</u>: The maximum combined pregnancy leave and parental leave to which an employee is entitled is fifty-two (52) weeks.
- (e) Interruption of leave by hospitalization of child: Notwithstanding Articles 29.10 (a), (b), (c) and (d), where an employee has begun parental leave pursuant to these Articles and the child to whom the parental leave relates is hospitalized for a period exceeding or likely to exceed one (1) week, the employee is entitled to return to and resume work in accordance with Article 29.10(n) and defer the unused portion of the parental leave until the child is discharged from the hospital, upon giving the Employer notice in accordance with Article 29.10 (g), (h) and (i).
- (f) <u>Limitation</u>: An employee is entitled pursuant to Article 29.10 (e) to only one (1) interruption and deferral of each parental leave.
- (g) Notice: An employee shall give the Employer four (4) weeks notice of
  - (i) the date the employee will begin parental leave pursuant to Article 29.10 (a): and
  - (ii) the date the employee will return to work upon completion of the leave unless the employee will take the maximum leave to which the employee is entitled
- (h) <u>Amendment of notice</u>: Notice given pursuant to Article 29.10 (g) may be amended from time to time by the employee
  - (i) by changing any date in the notice to an earlier date if the notice is amended at least four (4) weeks before that earlier date;
  - (ii) by changing any date in the notice to a later date if the notice is amended at least four (4) weeks before the original date; and
  - (iii) by adding the date that the employee will return to work if the notice is amended at least four (4) weeks before the employee would have been required to return to work.
- Short notice: An employee shall give the Employer as much notice as reasonably practicable of
  - (i) the return to work of the employee pursuant to Article 29.10 (e); and
  - (ii) the resumption of parental leave by the employee in accordance with Article 29.10 (e);

and Article 29.10 (g) does not apply.

- Notice in writing: Notice given pursuant to this Article shall be put in writing where the Employer so requests.
- (k) Proof of entitlement: Upon the request of the Employer, where an employee takes parental leave pursuant to Article 29.10 (a) (b) and (c), interrupts and defers leave pursuant to Article 29.10 (e) or gives notice pursuant to Article 29.10 (i), the employee shall provide such proof as is reasonably necessary to establish the entitlement of the employee pursuant to those provisions.
- (1) <u>Certificate as proof</u>: The certificate of a legally qualified medical practitioner is sufficient proof for the purpose of Article 29.10 (k) of the matters attested to in the certificate.
- (m) <u>Maintenance of benefit plan</u>: While an employee is on parental leave, the Employer shall maintain coverage for Group Life (Article 31.02), Extended Health Benefit Plan (Article 31.03), Weekly Indemnity and Long Term Disability (Article 31.04) and shall pay the premium cost for maintaining such coverage during the period of the parental leave.
- (n) Resumption of work: When an employee returns to work upon the expiry of the leave of absence taken pursuant to Article 29.10, the Employer shall permit the employee to resume work in the same position he/she held prior to the commencement of the parental leave, with no loss of seniority or benefits accrued to the commencement of the parental leave. Where the position no longer exists, the matter shall be referred to the Joint Committee on Labour Relations.
- (o) Service and seniority: While on parental leave, an employee shall continue to accrue and accumulate service and seniority credits for the duration of her leave, and her service and seniority shall be deemed to be continuous. However, service accumulated during parental leave shall not be used for the purposes of calculating vacation leave credits. Vacation leave credits will be prorated for the period of time the employee is not on such leave during the vacation year.

# 29.11 Adoption Leave

- (a) An employee, who has been employed by the Employer for at least one (1) year and who becomes, before or after this Article comes into force, a parent of one or more children through the placement of the child or children in the care of the employee for the purpose of adoption of the child or children pursuant to the law of the Province is entitled to an unpaid leave of absence of up to fifty-two (52) weeks upon giving the Employer notice of the date that the employee will begin the leave and the date that the employee will return to work as required by Article 29.11(f).
- (b) <u>Beginning and end of Adoption Leave</u>: Adoption leave pursuant to paragraph (a):
  - (i) begins on such date coinciding with the child or children first arriving in the employee's home; and

(ii) ends not later than fifty-two (52) weeks after the child or children first arrive in the employee's home;

as determined by the employee.

- (c) Interruption of leave by hospitalization of child: Notwithstanding Articles 29.11

  (a) and (b), where an employee has begun adoption leave pursuant to those Articles and the child to whom the parental leave relates is hospitalized for a period exceeding or likely to exceed one (1) week, the employee is entitled to return to and resume work in accordance with Article 29.11 (I) and defer the unused portion of the parental leave until the child is discharged from the hospital, upon giving the Employer notice in accordance with Article 29.11 (g).
- (d) <u>Limitation</u>: An employee is entitled pursuant to Article 29.11 (c) to only one (1) interruption and deferral of each adoption leave.
- (e) Notice: An employee shall give the Employer four (4) weeks notice of
  - (i) the date the employee will begin adoption leave pursuant to Article 29.11 (a); and
  - (ii) the date the employee will return to work upon completion of the leave unless the employee will take the maximum leave to which the employee is entitled.
- (f) <u>Amendment of notice</u>: Notice given pursuant to Article 29.11 (e) may be amended from time to time by the employee
  - (i) by changing any date in the notice to an earlier date if the notice is amended at least four (4) weeks before that earlier date;
  - (ii) by changing any date in the notice to a later date if the notice is amended at least four (4) weeks before the original date; and
  - (iii) by adding the date that the employee will return to work if the notice is amended at least four (4) weeks before the employee would have been required to return to work.
- (g) Short notice: An employee shall give the Employer as much notice as reasonably practicable of
  - (i) the first arrival of the child or children in the employee's home where that arrival is not anticipated or occurs sooner than reasonably expected;
  - (ii) the return to work of the employee pursuant to Article 29.11 (b); and
  - (iii) the resumption of adoption leave by the employee in accordance with Article 29.11(c);

and Article 29.11 (e) does not apply.

- (h) <u>Notice in writing</u>: Notice given pursuant to this Article shall be put in writing where the Employer so requests.
- (i) Proof of entitlement: Upon the request of the Employer, where an employee takes adoption leave pursuant to Articles 29.11 (a) and (b), interrupts and defers leave pursuant to Article 29.11 (c), the employee shall provide such proof as is reasonably necessary to establish the entitlement of the employee pursuant to those provisions.
- (j) <u>Certificate as proof</u>: The certificate of an official in the Department of Community Services with knowledge of the proposed adoption is sufficient proof for the purpose of Article 29.11 (i) of the matters attested to in the certificate.
- (k) <u>Maintenance of benefit plan</u>: While an employee is on adoption leave, the Employer shall maintain coverage for Group Life (Article 31.02), Extended Health Benefit Plan (Article 31.03), Weekly Indemnity and Long Term Disability (Article 31.04) and shall pay the premium cost for maintaining such coverage during the period of the adoption leave.
- (1) Resumption of work: When an employee returns to work upon the expiry of the leave of absence taken pursuant to Article 29.11 (e) or (f), the Employer shall permit the employee to resume work in the same position he/she held prior to the commencement of the adoption leave. Where the position no longer exists, the matter shall be referred to the Joint Committee on Labour Relations.
- (m) Vacation leave credits will be prorated for the period of time the employee is not on such leave during the vacation year.
- (n) Service and seniority: While on adoption leave, an employee shall continue to accrue and accumulate service and seniority credits for the duration of his/her leave, and his/her service and seniority shall be deemed to be continuous. However, service accumulated during adoption leave shall not be used for the purposes of calculating vacation leave credits. Vacation leave credits will be prorated for the period of time the employee is not on such leave during the vacation year.

# 29.12 Adoption Leave Allowance

- (a) An employee entitled to adoption leave under the provision of this Agreement, who provides the Employer with proof that he/she has applied for and is eligible to receive employment insurance (EI) benefits pursuant to the Employment Insurance Act, 1996, shall be paid an allowance in accordance with the Supplementary Unemployment Benefit (SUB) Plan.
- (b) In respect to the period of Adoption Leave, payments made according to the SUB Plan will consist of the following:
  - (i) where the employee is subject to a waiting period of two (2) weeks before receiving EI benefits, payments equivalent to seventy-five percent (75%)

of her weekly rate of pay for each week of the two (2) week waiting period, less any other earnings received by the employee during the benefit period; and

- (ii) up to a maximum of ten (10) additional weeks, payments equivalent to the difference between the weekly EI benefits the employee is eligible to receive and eighty percent (80%) of her weekly rate of pay, less any other earnings received by the employee during the benefit period which may result in a decrease in the EI benefits to which the employee would have been eligible if no other earnings had been received during the period.
- (c) For the purposes of this allowance, an employee's weekly rate of pay will be one-half the bi-weekly rate of pay to which the employee is entitled for her classification on the day immediately preceding the commencement of the adoption leave.

In the case of a part-time employee, such weekly rate of pay will be multiplied by the fraction obtained from dividing the employee's time worked (as defined for the purpose of accumulating service) averaged over the preceding twenty-six (26) weeks by the regularly scheduled full-time hours of work for the employee's classification

- (d) Where an employee becomes eligible for a salary increment or pay increase during the benefit period, payments under the **SUB** Plan will be adjusted accordingly.
- (e) The Employer will not reimburse the employee for any amount she is required to remit to Human Resources Development Canada where her annual income exceeds one and one-half (1-½) times the maximum yearly insurable earnings under the Employment Insurance Act.

# 29.13 Interview Leave

When an employee participates in a personnel selection process for a position with the Employer, including a promotion, and the employee's presence is required for purposes of the selection or promotion process including such further period as the Employer considers reasonable for the employee to travel to and from the place where his/her presence is required:

- (a) the employee shall be granted a leave of absence with pay for the required period; or
- (b) the Employer shall reschedule the time when the employee's presence is required; provided the employee requests a leave of absence of his/her supervisor as soon as possible after the requirement of his/her presence is known.

## 29.14 Bereavement Leave

(a) If a death occurs in the employee's immediate family when the employee is at work, the employee shall be granted leave with pay for the remainder of his/her scheduled shift. The employee shall also be granted seven (7) calendar days' leave of absence effective midnight following the death and shall be paid for all shifts the employee is scheduled to work during the seven (7) calendar day period.

In the event of the death of an employee's father, mother, spouse or child the leave shall be the greater of:

- (i) the seven (7) calendar day period referred to above; or
- (ii) a continuous period during which the employee's regular scheduled shifts, if he/she had been at work, would have totaled forty-eight (48) hours; e.g. four (4) twelve (12) hour shifts or two (2) twenty-four (24) hour shifts.

"Immediate Family" is defined as the employee's father, mother, guardian, brother, sister, spouse, child, father-in-law, mother-in-law, step-child or ward of the employee, grandparent or grandchild of the employee, and a relative permanently residing in the employee's household or with whom the employee permanently resides.

- (b) Every employee shall be entitled to leave with pay up to a maximum of one (1) day in the event of death of the employee's son-in-law, daughter-in-law, brother-in-law or sister-in-law, and may be granted up to two (2) days for travel for purposes of attending the funeral and shall be paid for those travel days which are not regularly scheduled days of rest.
- (c) Every employee shall be entitled to one (1) day leave without pay, for the purpose of attending the funeral of an employee's aunt or uncle, niece or nephew, or the grandparents of the spouse of the employee.
- (d) The above entitlement is subject to the proviso that proper notification is made to the Employer.
- (e) If an employee is on vacation or sick leave at the time of bereavement, the employee shall be granted bereavement leave and be credited the appropriate number of days to his/her vacation or sick leave credits.

## **ARTICLE 30 - VACATIONS**

### 30.01 Vacations

- (a) Employees will be entitled to vacation leave with pay based on continuous service as of March 31 of the previous vacation year on the following basis:
  - (i) less than one (1) year: three (3) weeks on a pro rata basis;

- (ii) one (1) year or more: three (3) weeks;
- (iii) five (5) years or more: four (4) weeks;
- (iv) sixteen (16) years or more: five (5) weeks;
- (v) twenty-five (25) years or more: six (6) weeks.
- (b) Weeks of vacation leave under paragraph (a) will be converted to hours on the basis of one (1) week being equivalent to the average number of hours in an employee's regularly scheduled work week.
  - Example #1: Assume a Full-Time employee had five (5) or more years of service as of March 31 and is therefore entitled to four (4) weeks' vacation. If his/her regularly scheduled work week averages forty-two (42) hours, he/she receives one hundred sixty-eight (168) hours' vacation.
  - Example #2: If a Part-Time employee is regularly scheduled to work an average of twenty-four (24) hours per week, one (1) week is converted to twenty-four (24) hours.
- (c) Employees who are on unpaid leave or are otherwise absent without pay for more than ten percent (10%) of the regular hours of work for the employee's location shall have their paid vacation leave pro rated, provided however they shall not receive less than the following vacation benefits:
  - vacation leave: employees with one (1) year or more service as of March
     31 shall receive two (2) weeks' vacation (with vacation pay to be determined under clause (ii)); and
  - (ii) vacation pay will not be less than four percent (4%) of wages received from the Employer as taxable income in the previous vacation year, exclusive of vacation pay.
- (d) An employee who has received, in accordance with a collective agreement with a predecessor employer, more vacation leave entitlement than provided for in paragraph (a) above, shall continue to be entitled to that vacation leave, provided, however, an employee shall not be entitled to more than the six (6) weeks' maximum vacation leave entitlement under paragraph (a) above.

It is understood that this modified "grandparenting" applies only to employees previously employed under the New Glasgow collective agreement where employees were entitled to ten (I0), fifteen (15) or twenty (20) working days' vacation leave depending on length of service. New Glasgow employees who have received more working days' vacation than they would be entitled to under paragraph (a) shall continue to receive the number of working days they received under the New Glasgow collective agreement, subject to the six (6) week maximum.

# 30.02 Vacation Year

The vacation year shall be from April 1 to March 31.

## 30.03 Vacation Requests

Employees who wish to exercise their seniority rights for vacation scheduling shall submit their preferred vacation requests for the following vacation year, in writing, to the Supervisor by February 1. The Employer shall make all reasonable efforts to post the vacation schedule by the end of the first full week in March.

# 30.04 <u>Vacation Scheduling</u>

The Employer shall make reasonable efforts to accommodate an employee's request for vacation. The following procedure shall be used in scheduling vacations for employees:

- (a) Vacations shall be scheduled by site. For this purpose Halifax-Dartmouth-Bedford-Sackville shall be considered one (1) site.
- (b) Subject to operational requirements (not ALS/BLS) preference for vacation at each site shall be on the basis of seniority, provided however:
  - (i) an employee shall not have preference based on seniority for more than two (2) weeks of vacation until other employees have had an opportunity to be scheduled for up to two (2) weeks;
  - (ii) an employee who fails to submit his/her vacation preference by February 1 shall lose any preferred status to other employees who have indicated their preference before that date. Vacation requests received after March 1 shall be considered on a first come first serve basis:
- (c) Employees will normally be required to take their vacation in blocks equivalent to the normal work cycle (for example, one (1) full week), provided however, employees may split their vacation periods into smaller increments (but no smaller than a full shift) with the approval of the Employer.

# 30.05 Payout of Vacation Entitlements

- (a) An employee terminating his/her employment at any time in his/her vacation year before he/she has had his/her vacation shall be entitled to a proportionate payment of salary or wages in lieu of such vacation.
- (b) An employee, upon his/her separation from the Employer, shall compensate the Employer for vacation which was taken but to which he/she was not entitled.
- (c) When the employment of an employee who has been granted more vacation with pay than he/she has earned is terminated by death, the employee is considered to have earned the amount of leave with pay granted to him/her.

#### 30.06 Illness During Vacation

When an employee is scheduled to commence vacation and becomes ill or injured immediately prior to the commencement of his/her vacation or is admitted into hospital during his/her vacation (including an employee who is admitted into hospital during his/her vacation and also discharged during his/her vacation), an equivalent number of sick days may be taken provided he/she has the necessary sick leave credits and the vacation days will be rescheduled at a later date agreed upon between the Employer and the employee. This provision applies only if the illness, injury or hospitalization can be established by the employee to the satisfaction of the Employer.

## 30.07 Bereavement Leave During Vacation

When an employee is entitled to paid bereavement leave during his/her period of vacation, an equivalent number of days may be taken as rescheduled vacation days at a later date to be agreed upon between the Employer and the employee.

## 30.08 Vacation Carry-Over

An employee shall be granted permission to carry over one week of his/her vacation leave each year to the following year.

## 30.09 Vacation Records

An employee is entitled to be informed, upon request, of the balance of his/her vacation leave with pay credits.

#### 30.10 Recall from Vacation

The Employer shall not recall an employee to duty after he/she has proceeded on vacation leave, subject to mutual agreement with the individual employee.

# ARTICLE 31 - EMPLOYEE BENEFITS

## 31.01 Pension

The Employer shall maintain a defined contribution pension plan for the benefit of employees. The Employer's contribution to the pension plan shall be 4% (increasing to 5% effective April 1, 2006 and to 6% effective April I, 2007) of an employee's earnings based on regular salary and the Employer's contribution shall be matched by each employee.

## 31.02 Group Life Insurance Plan

It is a condition of employment that employees covered by this Agreement will participate in the Group Life Plan for employees. Employees shall pay the full cost of the Group Life Insurance.

SMSSW1051363.15

## 31.03 Extended Health Benefit Plan (Medical and Dental)

It is a condition of employment that employees covered by this Agreement will participate in the Extended Health Benefit Plan for employees. The Employer shall pay the full cost of this plan.

## 31,04 Short Term Disability and Long Term Disability

- (a) It is a condition of employment that the employees covered by this Agreement will participate in the Long Term Disability Program for employees. Employees shall pay the full cost of this program.
- (b) It is a condition of employment that the employees covered by this Agreement will participate in the Short Term Disability (Weekly Indemnity) Program for employees. Employees shall pay the full cost of this program.

# 31.05 Regular Part-Time Employees

In order for a Regular Part-time employee to be eligible to participate and to continue to participate in the group insurance (Articles 31.02, 31.03 and 31.04), the employee must work at least twenty-four (24) hours per week, on average over the previous six (6) months.

# 31.06 Benefits Committee

Matters of general concern regarding employee benefits under Article 31 may be referred by either party to the Joint Committee on Labour Relations established pursuant to Article 21.01.

## 31.07 Emulover's Responsibility

The Employer's responsibility under this Article is limited to the payment of necessary premiums to purchase the required insurance. It has no liability for the failure or refusal of the insurance carrier to honour an employee's claim or to pay benefits and no such action on the part of the insurance carrier shall be attributable to the Employer or constitute a breach of this Agreement by the Employer. Under no circumstances shall the Employer be responsible for paying any benefits under this Article. No dispute arising under or relating to this Article shall be subject to the grievance and arbitration procedures set forth in Articles 11 and 12 of this Agreement, except an allegation that the Employer has failed to pay the premiums required to purchase the required insurance coverage.

# **ARTICLE 32 - SEVERANCE PAY**

# 32.01 Severance Pay

(a) An employee who is in receipt of layoff notice and who opts to resign with severance pay in accordance with Article 21.07 (d)(iii) shall receive severance

SMSSW1051363.15

pay in the amount equal to one week pay for each year of service to a maximum of twenty-two weeks pay.

(b) An employee who has his/her Long Term Disability benefits terminated after twenty-four (24) months, but before the expiry of thirty-six (36) months, shall be entitled to be placed on the recall list for twelve (12) months. If the employee is not placed in a position by the end of the twelve (12) months, he/she shall be entitled to receive severance pay as set out in (a) above.

## 32.02 Outstanding Debts Upon Death

If an employee whose employment is terminated by death has any outstanding debts with the Employer, the debts shall be deemed to have been paid in full.

## ARTICLE 33 - EMPLOYER LIABILITY

#### 33.01 Legal Expenses

The Employer shall pay an employee's legal costs up to a maximum of \$2,500.00 for any claim, action or other proceeding against the employee where the claim arises from the employee's work performance, provided:

- (a) the employee acted in good faith within the scope of his/her employment;
- (b) this does not apply to traffic violations by the employee unless the employee has been found not guilty, or to arbitration cases; and
- (c) this is not applicable where the employee is entitled to recover legal costs under any insurance policy.

#### ARTICLE 34 - AMENDMENT

34.01 This agreement may be amended in writing by the mutual consent of both parties.

#### **ARTICLE 35 - CORRESPONDENCE**

35.01 Any notice, in writing, that a party wishes to give to the other party, shall be given by mail, prepaid and addressed as follows:

To the Employer: Director of Human Resources

EMC Emergency Medical Care Inc.

Suite 300

239 Brownlow Avenue Dartmouth NS B3B 2B2

To the Union:

International Union of Operating Engineers,

Local 968B PO Box 700 28 Aberdeen Street Kentville NS B4N 3X9

Either party may change its address at any time by giving notice of such change to the other party, as above provided.

35.02 Employees shall keep their current residential address and telephone number on file with the Employer and where any employee moves during the currency of the Collective Agreement, such employee shall deliver to the Employer a written notice of such employee's new address and telephone number. The Employer shall be entitled to rely on the residential address and telephone number most recently on file with the Employer, when seeking to contact the employee. The Employer agrees not to release this information, without the consent of the employee, except to EHS or as required by law.

## ARTICLE 36 - TERM OF AGREEMENT

## 36.01 <u>Term</u>

This Agreement shall be in force from the 1st day of April, 2005 (subject to Article 36.02) and shall remain in force until the 31st day of March, 2008 and thereafter from year to year, unless one of the parties hereto notifies the other party in writing, within a period of not more than two (2) months prior to the scheduled termination date of the Collective Agreement, of its intention to revise and mend this Collective Agreement.

# 36.02 Retroactivity

Only the wage rates set out in Schedule "A" are retroactive and only employees on staff on the dale of signing of this Agreement and employees who have retired since April 1, 2005 are entitled to retroactive pay.

DATED at Halifax, Nova Scotia, this \_\_\_\_\_ day of June, 2007.

SIGNED ON BEHALF OF THE

Norman Earle

Dwayne Fitzgerald

SIGNED ON-BEHALF OF THE

George McLellan

Doug Sabean

Terry Chapman

Page 67
Susanne Dittmer

Brian Burkel

Gregory Haun

David LeCoar

Lecalus MacIntosh

Peter Maora

Terry Muise

M

- Page 68 -

# SCHEDULE A (SALARY SCALE)

	Classification		Step 1	Step 2	Step 3	Step4	Step 5
Apr. 1/05	PCP	Annual Hrly - 42	\$36,540 16.73	\$37,617 17.22	\$38,666 17.70	\$39,987 18.31	\$40,956 18.75
	ICP	Annual Hrly - 42	\$38,307 17.54	\$39,417 18.05	\$40,574 18.58	\$42,152 19.30	\$43,563 19.95
	ACP	Annual Hrly - 42	\$43,692 20.01	\$44,928 20.57	\$46,335 21.22	\$48,432 22.18	\$50,644 23.19
Apr. 1/06	PCP	Annual Hrly - 42	\$37,636 17.23	\$38,746 17.74	\$39,826 18.24	\$41,187 18.86	\$42,185 19.32
	ICP	Annual Hrly - 42	\$39,456 18.07	\$40,600 18.59	\$41,791 19.14	\$43,417 19.88	\$44,870 20.54
	ACP	Annual Hrly - 42	\$45,003 20.61	\$46,276 21.19	\$47,725 21.85	\$49,885 22.84	\$52,163 23.88
Apr. 1/07	PCP	Annual Hrly - 42	\$38,765 17.75	\$39,908 18.27	\$41,021 18.78	\$42,423 19.42	\$43,451 19.90
	ICP	Annual Hrly - 42	\$40,640 18.61	\$41,818 19.15	\$43,045 19.71	\$44,720 20.48	\$46,216 21.16
	ACP	Annual Hrly - 42	\$46,353 21.22	\$47,664 21.82	\$49,157 22.51	\$51,382 23.53	\$53,728 24.60

Note: annual rates are based on a regular work week of 42 hours per week on average.

## SCHEDULE B ADMINISTRATIVE AREAS

1. Administrative areas refer to the following groupings which are set out by region as follows:

# Cape Breton Region

- a) Baddeck, Neil's Harbour, Cheticamp, Inverness, Margaree
- b) Glace Bay, Sydney, North Sydney, New Waterford, Eskasoni
- c) St. Peters, Arichat, Port Hawkesbury

# Northeastern Region

- d) Amherst, Springhill, Parrsboro
- e) Pugwash, Oxford. Tatamagouche
- f) Truro, Milford
- g) Pictou County
- h) Antigonish, Guysborough
- i) Canso, Country Harbour, Sherbrooke
- j) Musquodoboit, Sheet Harbour

## **Central Region**

k) Halifax, Dartmouth, Bedford, Sackville

# Western Region

- 1) Chester, Hubbards, Tantallon, Mahone Bay, Lunenburg
- m) Bridgewater / New Germany, Liverpool
- n) Shelburne, Barrington, Woods Harbour
- o) Yarmouth, Meteghan, St. Bernard, Pubnico
- p) Berwick, Middleton, Bridgetown
- q) Digby / Long and Brier, Annapolis
- r) Kentville, Windsor, Wolfville

SMSS\\105 B63.15

- 2. The administrative areas listed in (a) through (r), and also the regions, are subject to change by the Employer, provided that the change will not be made on an arbitrary basis or without full consultation with the Union.
- 3. There may be an overlap between regions and administrative areas; for example, at one time, Antigonish was in an administrative area in the Northeastern Region, but was part of a different Region.

# MEMORANDIM OF AGREEMENT #1

Sings this is only the second province wide Collective Agreement between the parties and there have been difficulties establishing the propertient of the for some employees, the parties agree to the following:

The parties agree to establish a Joint Committee (consisting of two (2) representatives from each party) to review the work-history of employees who object to their date on the sentonity lieuraposted.

The Employers will provide the light Committee with work history that it has in its possession for each employee

An employed who requests a review of his/her work history shall be responsible for providing all relevant information to the Joint Committee.

After a claim has been reviewed by the Joint Committee, the Union shall make a recommendation to the Employer as to how the claim should be resolved.

II the Employer accepts the Union state employees.

If the Employer does not accept the Umon's recommendation, the dispute shall be referred to a single arbitrator in accordance with paragraphs 2 & 3 of this Memorandum.

- 2. The following expedited arbitration process will be used to resolve any dispute arising from an employes is objection to market date on the posted seniority list:
  - (a) the parties will meet to affempt to resolve the dispute;
  - (b) failing resolution, the padies will refer the dispute to a single arbitrator.
  - (c) If the dispute proceeds to arbitration, the parties shall make reasonable efforts to:
    - (1) establish and attempt to agree on the facts and prepare an Agreed : Statement of Facts (which may be supplemented by oral evidence);
    - (ii) forward to the other party, a copy of any document that it intends to use during any hearing and,
    - (iii) minimize the use of witnesses at any hearing:
  - (d) on ragiost of the parties, the arbitrator may, prior to rendering a decision, assist the parties in mediating a resolution of the grievance;
  - (e) falling resciution, the arbitrator shall be instructed to render a written decision to within seven (7) days of the presentation to him of an Agreed Statement of Facts of it the event of a hearing within seven (7) days of the hearing.

CEASCO INSTANCE

(f) the decision of the arbitrator shall be final and binding upon both parties.

3. No decision of the Employer based on seniority (other than a layoff) shall be reversed and the Employer shall not be liable for any such decision if such decision is based on the seniority list as posted unless the decision was made after the list had been corrected in accordance with an agreement of the parties or the decision of an arbitrator.

DATED at Halifax, Nova Scotla, this \_\_\_\_ day of June, 2007.

SIGNED ON BEHALF OF THE UNION	SIGNED ON BEHALF OF THE EMPLOYER
Norman Earle	Gaorge McLellan
Dwayne Herrald	Doug Sabran Julimer
Terry Champan	Susanne Differen
Briant Bunch	Barl Russell Kully Folly
Gregory Hann	Kelly Petroy
David LeGood	Grafiam McAHister
Lorslei MacIntosh	
Perer Moor	
Corry Mailso	$\gamma^o$

SM98V1051363.15

# MEMORANDUM OR AGREEMENT #2 OR AND PARENTHNG OF SURVIOLE GREDITS

The parties agree that any employes who, as of the date of signing, has received service credits will continue to receive such credits.

The Union agrees that the continuation of these service credits does not entitle any other employees who may not have received similar credits to receive such credits and does not constitute inconsistent application of discrimination which is contrary to the Agreement.

The Union agrees that it will not advance any grisvance, either a Union grievance or an individual grisvance, which is based on the continuation of such credits or alleges that the Employer has been inconsistent in its application of Article 18:04 or that the Employer has discriminated egainst other employees by grandparenting the aforementioned employees.

DATED at Halifax, Nova Scotta, this II day of June, 2007.

SIGNED ON BEHALF OF THE UNION	SIGNED ON BEHALF OF THE EMPLOYED
Norman Berle	Georga McLellan
Dwayne till begorald  Terry Changan	Susquise Littuses Susanne Dittmer
Brian Bunch	Earl Russell Kolly Folly
Gregory Hanny David Leggdr	Kelly Epley!
Gorgia Wacintosh	
Parcy Marie	
2M84/1031340214	

# MEMORANDUM OF AGREEMENT #3 (POLICY OF SHIFT SWITCHES)

The Employer shall establish a province-wide policy in regard to when shift switches will be allowed. (For example, a shift switch shall not be allowed where it would decrease the number of ALS crews operating at any time.)

The terms and administration of this policy on shift switches shall not be subject to the grievance and arbitration provisions of the Collective Agreement between the parties. However, matters of general concent regarding this policy (but not complaints with respect to individual circumstances) may be referred by the Union to the Joint Committee on Labour Relations established pursuant to Article 21 of

day of June, 2007.

SIGNED ON BEHALF OF THE UNION BEHALF OF THE BMPLOYER Dong Setsean

Dong Setsean

Ferry Chapman

Brian Binch

Gregory Figure

Kelly Eoley

Kelly Eoley

DATED at Halifax; Nova Scotla, this <u></u>

SMSS41051363.J5

# [to be prepared on emcletterhead]

(Dais): June: II, 2007 \$0 Mr. Norman Parl Thiomational Union of Operating Engineers, Local 968B

PO Box 700 Se Abardeen Street Kentylle NS: BAN 3X9

Dear Norm:

This letter is further to Hiscussions during associations with respect to the rate of pay for paramedies who have officially scarce (that is, eligible for Canada Pension Plan) from EMC, but commune to work for the Company as a Casual.

The Company's practice is to pay such persons no less than the actual rate of pay which they were receiving at his time of retirement.

This letter does not form part of the Collective Agreement, nor does it in any way give fights to tetrees; but is for your information only.

Yours truly,

, Susanne Dittmer Director, Huttein Rescurces

SM88W(051368:14

# [TO BE PREPARED ON EMC LETTERHEAD]

May = 2007 June 11, 2007

Mr. Norman Earl International Union of Operating Engineers, Local 968B PO Box 700 28 Aberdeen Street Kentville NS B4N 3X9

Dear Norm:

This letter - which does not form part of the Collective Agreement - is further to discussions during negotiations in regard to situations in which only one paramedic is assigned to an ambulance.

As discussed, EMC's practice is to schedule two paramedics per ambulance and if a paramedic is absent, EMC makes every reasonable effort to obtain a replacement, including offering the shift to employees at time and one-half pursuant to the second round of the call-in system referred to in Article 28.02 (c), so as to avoid a situation where only one paramedic is assigned to an ambulance. Should that situation urise, the ambulance with only one paramedic is used as a "back-up" vehicle in that, to the extent practical, other ambulances will be assigned to a call before the ambulance with only one paramedic.

EMC recognizes that there are limitations on the services that can be provided by one paramedic on his/her own, rather than as a member of a two person team, but patient care dictates that it is better to have an ambulance with one paramedic than to take the ambulance out of service. If the situation arises where only one paramedic is assigned to an ambulance, I would be pleased to review the circumstances with you.

Yours truly.

S. Dittuer

Susanne Dittmer Director, Human Resources 8)